

Title	Operational Design for High-density Culture of Human Induced Pluripotent Stem Cells in Suspension
Author(s)	Nath, Suman Chandra
Citation	大阪大学, 2017, 博士論文
Version Type	VoR
URL	https://doi.org/10.18910/67132
rights	
Note	

The University of Osaka Institutional Knowledge Archive : OUKA

https://ir.library.osaka-u.ac.jp/

The University of Osaka

## Operational Design for High-density Culture of Human Induced Pluripotent Stem Cells in Suspension

Nath Suman Chandra July, 2017

Graduate School of Engineering Osaka University

## Operational Design for High-density Culture of Human Induced Pluripotent Stem Cells in Suspension

A dissertation submitted to the Graduate School of Engineering Osaka University in fulfillment of the requirements for the degree of Doctor of Philosophy in Engineering

> By Nath Suman Chandra July, 2017

#### Preface

This study was conducted under the supervision of Professor Masahiro Kino-oka at the Department of Biotechnology, Graduate School of Engineering, Osaka University, Japan from 2012 to 2017.

The main objective of this thesis is to improve the operational design of human induced pluripotent stem cells (hiPSCs) culture for obtaining high-density of hiPSCs through the simplification of culture operations, as well as low supplementation of growth factors to promote a cost-effective platform in suspension. The most interesting focus was the investigating the growth properties of hiPSCs to set up a boundary condition for preferential growth in single aggregate culture, revealing the temporary effect of botulinum hemagglutinin (HA) for disrupting E-cadherin mediated cell-cell connections for the establishment of an *in situ* simple method of hiPSC aggregate break-up in bioreactor for obtaining high-density, and the establishment of a culture medium refinement method by dialysis for recycling autocrine factors, as well as reducing the consumption of cytokines for the expansion of hiPSCs in suspension culture. The author believes that the findings obtained in this study will lead to the establishment of an automatic, closed, and cost-effective platform for large-scale expansion of hiPSCs in suspension culture.

Nath Suman Chandra Department of Biotechnology Graduate School of Engineering Osaka University Suita, Osaka 565-0871, Japan

1

### Contents

Abstract	5
General introduction	8
Chapter 1: Size- and time-dependent growth properties of human induced pluripotent	
stem cells in the culture of single aggregate	15
1.1 Introduction	15
1.2 Materials and methods	10
1.2 Materials and methods	10
1.2.1 Cells and culture conditions	16
1.2.2 Culture of single hiPSC aggregates	17
1.2.3 Determination of cell number, specific growth rate and aggregate size	18
1.2.4 Fluorescent staining	19
1.2. 5 Preparation of conditioned medium	19
1.2.6 Statistical analysis	20
1.3 Results	21
1.4 Discussion	27
1.5 Summary	32
Chapter 2: Botulinum hemagglutinin-mediated in situ break-up of human induced	
pluripotent stem cell aggregates for high-density suspension culture	33
2.1 Introduction	33
2.2 Materials and methods	35
2.2.1 Cells and Culture Conditions	35
2.2.2 HA Addition Conditions for Aggregate Break-up	35
2.2.3 Suspension Culture of hiPSCs	36
2.2.4 Kinetic Analysis of Growth Properties of hiPSC Aggregates	38
2.2.5 Analysis of Medium Components	38
2.2.6 Flow cytometry	39
2.2.7 Fluorescent Staining	39
2.2.8 Statistical analysis	40

2.3 Results	40
2.3.1 Break-up of hiPSC Aggregates into Small Sizes with HA	40
2.3.2 Effect of break-up of aggregates by HA on hiPSCs growth	43
2.3.3 High-density Culture of hiPSC Aggregate with HA addition	44
2.4 Discussion	47
2.5 Summary	54
Chapter 3: Culture medium refinement by dialysis for the expansion of human induced	
pluripotent stem cells in suspension	55
3.1 Introduction	55
3.2 Materials and methods	56
3.2.1 Culture of hiPSCs	56
3.2.2 Effect of lactic acid	57
3.2.3 Medium refining system	58
3.2.4 Adherent culture of hiPSCs with refined medium	59
3.2.5 Suspension culture with the medium refining system	59
3.2.6 Flow cytometry	60
3.2.7 Stabilization of bFGF	60
3.2.8 Statistical analysis	61
3.3 Results	61
3.3.1 Effect of lactic acid on hiPSC growth and removal of lactic acid by dialysis	61
3.3.2 Cytokines in the refined medium	62
3.3.3 Adherent culture with refined medium	64
3.3.4 Suspension culture with the medium refining system	65
3.4 Discussion	68
3.5 Summary	74

Chapter 4: Suspension culture of hiPSCs with medium refinement and	aggregate break-up
for high-density and high-yield	75
4.1 Introduction	75

4.2 Materials and methods	77
4.2.1 Culture of hiPSCs	77
4.2.2 Culture with medium refinement system	77
4.2.3 Suspension culture with medium refinement and aggregate break-up	78
4.2.4 Kinetic Analysis of Growth Properties of hiPSC Aggregates	79
4.2.5 Analysis of Medium Components	80
4.2.6 Flow cytometry	80
4.2.7 Fluorescent Staining	81
4.2.8 Statistical analysis	82
4.3 Results	82
4.4 Discussion	87
4.5 Summary	93
General conclusion	94
Proposals for future work	97
Nomenclature	99
Abbreviations	100
Literature Cited	101
List of publications	113
Acknowledgements	114

#### Abstract

Large numbers of human induced pluripotent stem cells (hiPSCs) are required for cell therapy applications. Suspension culture of hiPSCs is superior to obtain large numbers of cells in comparison to adherent culture since high cell number  $(10^8-10^9 \text{ cells})$ , as well as better control of process parameters can be obtained in stirred suspension bioreactor. In the conventional suspension culture, hiPSCs show initial cell loss during seeding due to single cell dissociation, low growth rate due to aggregate heterogeneity, and size limitation during late-stage of culture which results in low cell density. Moreover, toxic metabolites (lactic acid, ammonium, etc.) accumulate in the culture medium which requires frequent medium change which along with remove autocrine factors from medium in conventional suspension culture medium also results in low cell density in suspension culture. By considering these limitations, this study aimed at establishing a simple and scalable method by improving the operational design of hiPSC culture for obtaining high cell density with efficient recycling of autocrine factors in suspension.

In chapter 1, size- and time-dependent growth properties of hiPSC aggregate was investigated for setting up a boundary condition in which hiPSCs can maintain high growth rate in suspension culture. hiPSCs showed size-dependent growth heterogeneity during early-stage, and time-dependent collagen type I accumulation on the aggregate periphery during late-stage of culture. Thus, considerations for initial cell number and culture time were important to maintain the balance between the secretion of autocrine factors, and the accumulation of extracellular matrix on the aggregate periphery to achieve preferable growth of hiPSCs in single aggregate culture. In chapter 2, a simple method for *in situ* break-up of hiPSC aggregates was established by revealing the temporary effect of botulinum hemagglutinin (HA) for the disruption of E-cadherin mediated cell-cell connections in suspension culture. This method allowed high-density culture of hiPSCs through the break-up of aggregates into small sizes without initial cell loss during seeding in suspension culture. It also provided the simplification of culture operations without enzymatic treatment or washing in suspension culture.

In chapter 3, a culture medium refinement method was established by using a dialysis filter by which toxic metabolites (lactic acid, ammonium, etc.) was removed from the medium, and bFGF, TGF- $\beta$ 1, and other autocrine factors were recycled efficiently for the expansion of hiPSCs in suspension culture. This method allowed expansion of hiPSCs by changing the medium only once and recycling growth factors, as well as other autocrine factors that promoted a costeffective platform in suspension culture.

In chapter 4, medium refinement system was combined with aggregate break-up by HA during late-stage of culture for obtaining high density of hiPSCs in suspension. By using this system, higher growth rate as well as higher cell density was obtained by a minimum supplementation of growth factors which provided 21-times higher yield on culture medium in comparison to the conventional suspension culture.

Overall, a simple and scalable method for high-density culture of hiPSCs was established by setting up a boundary condition for maintaining high growth rate, and breaking up the aggregates into small sizes with HA in suspension culture. This method also provided the simplification of culture operations without enzymatic treatment or washing in suspension. Moreover, the integration of medium refinement system with aggregate break-up by HA provided high-density of hiPSCs with high-yield on culture medium by efficient recycling of autocrine factors in suspension culture. Thus, the improved operational design of hiPSC culture provided a cost-effective platform for obtaining high cell density with the simplification of culture operations which will promote simple closed-system for large-scale expansion of hiPSCs in suspension culture.

#### **General introduction**

After the first isolation of human induced pluripotent stem cells (hiPSCs) by Yamanaka group (Takahashi *et al.*, 2007), application of these cells are increasing day by day as regenerative medicine for cell replacement therapies (Schulz *et al.*, 2012), tissue engineering (Badylak *et al.*, 2011), and drug discovery (Wernig *et al.*, 2007) due to their pluripotency and self-renewal capacity. hiPSCs are reprogrammed from skin cells of similar human leukocytes antigen (HLA)-type healthy donors and stored in master cell bank (**Fig. 1**). From the master cell bank, cells are collected and expanded to make working cell bank. After expansion of hiPSCs from working cell bank, cells are differentiated into desired cell type, and transplanted into the patient as a regenerative medicine. To realize their potentiality, large-scale expansion of hiPSCs is important as practically more than  $10^9$ – $10^{10}$  cells are required for cell therapy applications (Lock and Tzanakakis, 2007; Jing *et al.*, 2008; Serra *et al.*, 2010; Kehoe *et al.*, 2010; Olmer, *et al.*, 2010).



Fig. 1 Schematic diagram showing the application of hiPSCs as regenerative medicine.

Before starting large-scale expansion of hiPSC in suspension culture, some important points need to be considered, especially operational techniques (culture system, medium feeding regimen, target final cell density, etc), economic feasibility (culture medium), quality and safety of the final harvest are the most important considerations for using hiPSCs in clinical applications (Jenkins and Farid, 2015). To date, there are two main systems for hiPSCs expansion: adherent (2D) and suspension (3D) culture.

Characteristics	Adherent Culture (2D)	Suspension culture (3D)
Final cell density	Low	High
Medium consumption	High	Low
Growth surface area	Limited	Not limited
Required culture vessels	High numbers	Single bioreactor
Online sampling	Not possible	Possible
Labor	Intensive	Flexible
Scalability	Low	High
Cell harvesting	Difficult (need to detach from culture surface)	Easy (ready to use as aggregate)

Table 1 Comparison of hiPSC culture between adherent (2D) and suspension (3D).

Expansion of hiPSCs is easier in adherent culture in comparison to suspension culture, and large-scale expansion process has already been established by using automatic system, especially in Cell-STAK<sup>TM</sup>, Nunc cell factory<sup>TM</sup>, CompacT SelecT<sup>TM</sup>, as well as by using many other systems. The major limitation in adherent culture is the growth surface area where high cell density of hiPSCs is difficult to obtain. Moreover, culture cost is high due to high consumption of medium in adherent culture (**Table 1**). On the other hand, large-scale expansion of hiPSC in suspension culture is possible as high number of cells can be obtained with low consumption of medium by using a single bioreactor (Zweigerdt *et al.*, 2011; Olmer *et al.*, 2012).

Although hiPSC can be cultured by many methods in suspension: aggregate, encapsulation, and microcarrier culture, aggregate culture is promising due to self-formation of aggregates from single cells and aggregates are easy to harvest from suspension culture. During initial seeding in suspension culture, single cells are inoculated into the culture vessel, and cells agglomerate to form sphere-shaped aggregates by encountering and connecting with the neighboring cells which is mediated by E-cadherin (Li et al., 2012; Li et al., 2014) (Fig. 2). The size of aggregates increases with the increase in culture time and some of the aggregates spontaneously fuse to form an un-uniform shape (Bauwens et al., 2008; Otsuji *et al.*, 2014), however, some of the aggregates collapse due to high shear stress. The prolongation of culture time leads to undesired differentiation, as well as cell necrosis inside the aggregate which also cause the difficulty in maintaining proper growth in suspension culture (Sachlos et al., 2008). To overcome these difficulties, Singh et al. performed passage culture by dissociating the aggregates into single cells with Rhoassociated protein kinase (Rock) inhibitor (Watanabe et al., 2007), but it could not prevent initial cell loss during seeding in suspension culture (Singh et al., 2010).



Fig. 2 Schematic illustration showing the growth properties of hiPSC aggregate in suspension culture.



Fig. 3 Comparison of hiPSCs growth rate and final cell density in suspension culture with other culture systems.

To prevent single cell apoptosis, several techniques were applied for passaging hPSC colony as small clumps by using divalent cations (Ohnuma *et al.*, 2014), or EDTA (Beers *et al.*, 2012), or sodium citrate (Nie *et al.*, 2014) in adherent culture. hiPSC colony can also be passaged by physically cutting into small pieces, and seeding into suspension culture (Son *et al.*, 2011). Although many protocols have already been established to expand hiPSC in aggregate suspension culture by seeding single cells with Rock inhibitor in aggregate suspension culture (Watanabe et al, 2007), still there is no established method for passaging hiPSC aggregates into small clamps in suspension culture (**Fig. 3**). The highest cell density obtained in these studies is ranging from  $1.0 - 2.0 \times 10^6$  cells mL<sup>-1</sup> by frequent changing of medium (Olmer *et al.*, 2012; Wu *et al.*, 2014). However, accumulation of toxic metabolites increases with the increase of cell density in suspension culture which inhibits the growth of hPSCs (Chen *et al.*, 2010). Although these components can be removed by perfusion culture, and hiPSCs can be expanded to a high

density  $(3.6 \times 10^6 \text{ cell mL}^{-1})$  (Kropp *et al.*, 2016), controlling aggregate size is a big challenge during the long-term suspension culture.



Fig. 4 Major drawbacks of hiPSC aggregate culture in suspension.

The major drawbacks in conventional suspension culture of hiPSCs are: (1) low growth rate due to aggregate heterogeneity, (2) dissociation of aggregates into single cells by enzymatic digestion which results in massive cell loss during seeding, (3) high culture cost as large volume of culture medium is required, and (4) low final cell density due to aggregate size limitation during late-stage of culture (**Fig. 4**). Aggregate size limitation as well as extracellular matrix (ECM)-shell formation hinders the diffusive transport of inductive biochemicals to and from aggregates which triggers the undesired differentiation inside the aggregate causing difficulty in maintaining proper growth of hPSCs in suspension culture. Culture cost depends on various factors, especially medium, process efficiencies, technology choices and resources. Culture cost also depends on growth rate ( $\mu$ ) as well as on final cell density ( $X_f$ ), because prolonged culture period consumes high volume of medium in suspension culture.

By considering these limitations in conventional suspension culture, therefore, an improved operational design for hiPSC culture was proposed for obtaining high cell density with low supplementation of growth factors in stirred suspension bioreactor (**Fig. 5**). To obtain high density of hiPSCs in suspension culture, the following four chapters were considered.

Chapter 1 describes the effect of aggregate size and culture time on the growth of hiPSCs in single aggregate culture. To evaluate the growth properties of single aggregate of hiPSC, the effect of initial cell number and culture time were investigated quantitatively, and the proliferative ability of cells in the aggregates was elucidated by immunohistological analysis. A boundary condition for preferable growth of hiPSCs was proposed for obtaining high growth rate in single aggregate culture.

Chapter 2 describes the establishment of a simple and scalable method for the expansion of hiPSCs by breaking up the aggregates into small sizes with botulinum hemagglutinin (HA), and elucidation of mechanisms of aggregate break-up in suspension culture. This method allowed high-density culture of hiPSCs through the simplification of culture operations without the need of enzymatic treatment or washing in suspension culture.

Chapter 3 focuses on the establishment of a culture medium refinement system by using a dialysis filter. This method allowed the removal of toxic metabolites, as well as recycling of autocrine factors in the culture medium, and reduced the use of macromolecules (transforming growth factor-beta 1 (TGF- $\beta$ 1), insulin, etc.) for the expansion of hiPSCs in suspension culture. The minimum inhibitory level of lactic acid to hiPSCs, and lactic acid removal by dialysis were also investigated in this study.



Stirred suspension bioreactor

**Fig. 5** Schematic diagram showing the strategy for obtaining high cell density in suspension culture by (1) analysis of growth properties of aggregate, (2) aggregate break-up to small sizes, (3) refinement of culture medium by dialysis, and (4) integration of medium refinement with aggregate break-up in suspension culture.

Chapter 4 focuses on the integration of Chapters 2 and 3 to obtain high cell density by breaking up the aggregates into small sizes, and refinement of culture medium for efficient recycling of autocrine factors, and other growth factors to reduce the supplementation of costlier macromolecules in suspension culture. The major growth-limiting factors during late-stage of culture were also investigated in this study.

#### Chapter 1

# Size- and time-dependent growth properties of human induced pluripotent stem cells in the culture of single aggregate

#### **1.1 Introduction**

Human induced pluripotent stem cells (hiPSCs) are considered as a promising cell source for cell therapy applications owing to their self-renewal capability and differentiation potential (Takahashi *et al.*, 2007; Kimbrel and Lonza, 2015). High cell numbers are generally required for clinical application, requiring robust methods for cell expansion (Kehoe *et al.*, 2010; Serra *et al.*, 2010). Conventional suspension culture of hiPSC aggregates has become a reliable method to obtain high numbers of cells  $(10^8 - 10^9 \text{ cells})$  (Olmer *et al.*, 2010; Zweigerdt *et al.*, 2011; Olmer *et al.*, 2012), in particular after seeding single cells with ROCK inhibitor which prevents single cell apoptosis (Watanabe *et al.*, 2007). During initial seeding in suspension culture, single hiPSCs are inoculated at a desired density into the culture vessel where the cells agglomerate to form sphere-shaped aggregates by encountering and connecting with neighboring cells, a process mediated by E-cadherin (Sugawara *et al.*, 2010; Li *et al.*, 2012).

Aggregate size increases with the increase in culture time, and coalescence occurs between aggregates to form larger ones leading to reduced proliferation in suspension culture (Otsuji *et al.*, 2014; Bauwens *et al.*, 2008; Abbasalizadeh *et al.*, 2012). The heterogeneous size of aggregates causes low growth efficiency since micro-environmental stimuli, such as, cell–cell connections and cell–soluble factor interactions are dependent on the aggregate size (Moon *et al.*, 2011). For example, oxygen concentration in the center region of large size aggregates (400 µm radius) was reported to reduce 50% in comparison to that in small sizes of aggregates (200 µm radius) of hPSCs (Kinney *et al.*, 2011; Winkle *et al.*, 2012; Wu *et al.*, 2014). Moreover, the size

of individual aggregate has been reported to influence the growth and differentiation of hPSCs in aggregate suspension culture (Burridge *et al.*, 2007; Hwang *et al.*, 2009; Mohr *et al.*, 2010; Moon *et al.*, 2014). Therefore, determining the optimum size of aggregate is important for obtaining high proliferation rate of hiPSCs in suspension culture. Moreover, prolonged culture period leads to low inward diffusion of oxygen and nutrients from the surface to the densely agglomerated cells in the center causing necrosis inside the aggregate (Novosel *et al.*, 2011; Jain *et al.*, 2005). Due to aggregate size limitation, as well as ECM-shell formation during longer period of culture, the diffusive transport of inductive biochemicals to and from aggregates are affected which triggers the undesired differentiation inside the aggregate causing difficulty in maintaining proper growth of hPSCs in suspension culture (Sachlos *et al.*, 2008). Consequently, detailed analysis of hiPSC aggregates is necessary to determine the boundary conditions for aggregate size and culture time that enable aggregates to retain a high proliferation rate and undifferentiated state in suspension culture.

In this study, to evaluate the growth properties of single aggregate of hiPSC, the effect of initial cell number and culture time were investigated quantitatively, and the proliferative ability of cells in the aggregates was elucidated by immunohistological analysis.

#### **1.2 Materials and Methods**

#### 1.2.1 Cells and culture conditions

The hiPSC line, Tic, was provided by the Japanese Collection of Research Bioresources (JCRB1331, JCRB Cell Bank, Japan). Cells were routinely maintained on polystyrene substrate coated with recombinant laminin-511 E8 fragments (iMatrix<sup>™</sup>-511; Nippi Inc., Japan) in commercially available medium (mTeSR<sup>™</sup>1; STEMCELL Technologies, Canada). For

subculture, single cells were seeded with 10  $\mu$ M ROCK inhibitor (Y-27632; Wako Pure Chemical Industries, Japan). Initial seeding was fixed at a viable cell density of  $1.0 \times 10^4$  cells cm<sup>-2</sup>. Cells were incubated at 37°C in a humidified atmosphere with 5% CO<sub>2</sub>, and medium was exchanged daily with fresh medium. On day 4, when cells reached 80–90% confluence, cells were subcultured. Briefly, hiPSCs were treated with 5 mM ethylenediaminetetraacetic acid (EDTA)/phosphate-buffered-saline (PBS) with 10  $\mu$ M ROCK inhibitor for 7 min at room temperature. Dissociation reagent (TrypLE Select<sup>TM</sup>, Invitrogen, USA) with 10  $\mu$ M ROCK inhibitor was then applied for another 7 min at room temperature. After dissociating the hiPSC colonies into single cells by pipetting (Miyazaki *et al.*, 2012), cells were re-seeded into a new culture dish.

#### **1.2.2 Culture of single hiPSC aggregates**

For single aggregate culture, hiPSC colonies were dissociated into single cells after treating with EDTA/PBS and TrypLE Select with 10  $\mu$ M Rock inhibitor as described above. After centrifugation, single hiPSCs were re-suspended in fresh medium. Viable cells were counted with a cell counter (TC20, Bio-Rad Inc., USA) by the trypan blue exclusion method. Single aggregate culture was carried out in 96-well ultra-low attachment V-bottom plates (Sumitomo Bakelite Co. Ltd., Japan), by seeding,  $N_0 = 4.0 \times 10^1$  to  $5.1 \times 10^3$  cells/well in 100  $\mu$ L of medium/well, and cultured for 120 h (**Fig. 1.1**). Aggregates containing small number of initial cells (especially for  $N_0 = 4.0 \times 10^1$ ,  $8 \times 10^1$ ,  $1.6 \times 10^2$ , and  $3.2 \times 10^2$  cells/aggregate) were cultured in multiple 96-well V-bottom plates to obtain sufficient number of cells (minimum density:  $5.0 \times 10^4$  cells/mL) for counting by the automatic cell counter. Half of the spent medium was changed with fresh medium at 48 and 96 h.

#### 1.2.3 Determining cell number, specific growth rate and aggregate size

After collecting hiPSC aggregates from the V-bottom plates at t = 24, 72, and 120 h, aggregates were washed with PBS, and dissociated into single cells using TrypLE Select with 10  $\mu$ M ROCK inhibitor. After centrifugation, supernatant was discarded and cells were resuspended in fresh medium. Numbers of viable cells in aggregates were counted using the cell counter, and cell number in a single aggregate, N (cells/aggregate) was calculated using the following equation: N = Total number of viable cells/Number of aggregates in the V-bottom plates. The values of N at t = 24, 48, 72, and 120 h denoted as  $N_{24}$ ,  $N_{48}$ ,  $N_{72}$ , and  $N_{120}$ , respectively.



Fig. 1.1 Schematic diagram showing the experimental procedures of the present study.

Live cell ratio,  $\alpha$  (-), was calculated using the following equation:  $\alpha = N_{24}/N_0$ . The apparent specific growth rate,  $\mu^{app}$  (h<sup>-1</sup>), was calculated using the following equation:  $\mu^{app} = \ln (N_{72}/N_{24})/(\Delta t)$  at t = 24 to 72 h, and  $\mu^{app} = \ln (N_{120}/N_{72})/(\Delta t)$  at t = 72 to 120 h, where  $\Delta t$  is the differential time of 48 h. To determine the diameter of aggregate, D ( $\mu$ m), bright-field images of hiPSC aggregates were captured at t = 24 h by using an image analyzer (InCell Analyzer 2000;

GE Healthcare, USA) with a  $10 \times$  objective lens, and *D* was determined by using an image processing software (Image-Pro Plus, Media Cybernetics Inc., USA).

#### **1.2.4 Fluorescent staining**

hiPSC aggregates were washed with PBS (Sigma-Aldrich, USA) and embedded in optical cutting temperature compound (OCT, Tissue-Tek, Japan). Thin sections (20 μm) were prepared using a cryostat microtome (Leica, Germany), and fixed with 4% paraformaldehyde (Wako Pure Chemical Industries) for 10 min at room temperature. After washing with PBS, the specimens were permeabilized with PBS containing 0.5% Triton X-100 (Wako Pure Chemical Industries) for 5 min, washed twice with PBS, and blocked in Block Ace (DS Pharma Co. Ltd., Japan) at 4°C overnight. The sections were then probed with primary antibodies against ki-67 (Abcam, USA), and collagen type I (Abcam) at 4°C overnight. Sections were then washed twice with Tris-buffered saline (TBS: Dako, Japan), and immersed in PBS containing 10% Block Ace and Alexa Fluor 488-conjugated secondary antibody (Life technologies, USA) for 60 min at room temperature. Cell nuclei were stained with 4′, 6-diamidino-2-phenylindole (DAPI, Life Technologies, USA) for 20 min. The specimens were washed with PBS, and observed with a confocal laser scanning microscope (FV1000, Olympus, Japan).

#### **1.2.5 Preparation of conditioned medium**

To investigate the effect of autocrine factors on small size aggregates, used medium was collected from aggregate culture  $((2.7 \pm 0.2) \times 10^2 \text{ cells/aggregate})$  in the V-bottom plates after 72 h, and concentrated using a dialysis system with a molecular weight cut off (MWCO) of 5 kDa (Vivaspin 20, Vivaproducts, USA). This method diluted the low molecular weight toxic

components from the collected medium (lactic acid, ammonium), and retained the high molecular weight components (cytokines) and autocrine factors. The resulting medium was then mixed with an equal volume of fresh mTeSR<sup>TM</sup>1 medium to prepare the conditioned medium (CM). Aggregates containing  $(2.8 \pm 0.5) \times 10^1$  and  $(8.8 \pm 0.7) \times 10^1$  cells/aggregate were cultured for 72 h with fresh medium and CM in 96-well V-bottom plates. Aggregates were harvested at t = 24 and 72 h, and dissociated into single cells using TrypLE Select. After counting the cells with a cell counter, the apparent specific growth rates were determined.

#### 1.2.6 Statistical analysis

Data presented in this study were obtained from three independent experiments performed in 96-well ultra-low attachment V-bottom plates and are expressed as means  $\pm$  standard deviation (SD). Statistical comparisons were evaluated using the Student's *t*-test and values of *p* < 0.01 and *p* < 0.05 were considered significant.

#### **1.3 Results**

To investigate the effect of aggregate size on hiPSC growth, aggregates containing different initial cell numbers were cultured for t = 72 h, and evaluated quantitatively. With increased  $N_0$ ,  $\alpha$  also increased, and the maximum  $\alpha = 0.7 \pm 0.1$  was observed for  $N_0 = 6.4 \times 10^2$  cells/aggregate, whereas, an  $N_0$  higher than this, resulted in decreased  $\alpha$  (**Fig. 1.2A**). However, *D* increased with the increase of  $N_{24}$  (**Fig. 1.2B**).



Fig. 1.2 Live cell ratio (A), and aggregate size (B) after 24 h of seeding.

Aggregates containing  $N_{24} = (8.8 \pm 0.8) \times 10^1$  to  $(2.7 \pm 0.2) \times 10^2$  cells showed higher cell expansion ratio after culturing for 120 h (**Fig. 1.3A**). Initial aggregate size played a major role in hiPSC growth, whereby  $\mu^{app}$  increased with increased  $N_{24}$ , and maximum  $\mu^{app}$  was observed in medium size aggregates ( $(8.8 \pm 0.8) \times 10^1$  to  $(6.8 \pm 1.1) \times 10^2$  cells/aggregate) during early-stage of culture (**Fig. 1.3B**). Small ( $(1.1 \pm 0.4) \times 10^1$  to  $(2.8 \pm 0.5) \times 10^1$  cells/aggregate), and large ( $(1.1 \pm 0.2) \times 10^3$  to  $(3.5 \pm 1.1) \times 10^3$  cells/aggregate) size aggregates showed a lower  $\mu^{app}$  in comparison to medium size aggregates during t = 24 to 72 h. However, small size aggregates containing  $N_{24} = (2.8 \pm 0.5) \times 10^1$  cells/aggregate showed higher  $\mu^{app}$  in CM (( $(3.8 \pm 0.3) \times 10^{-2}$  h<sup>-2</sup>

<sup>1</sup>) in comparison to fresh medium ((3.1 $\pm$ 0.2)  $\times$  10<sup>-2</sup> h<sup>-1</sup>) after culturing for 72 h which was similar to the  $\mu^{app}$  of medium size aggregates containing  $N_{24} = (8.8 \pm$  $(0.7) \times 10^1$  cells/aggregate (Fig. 1.4). To evaluate the effect of culture time on hiPSC growth, aggregates containing different  $N_0$  were cultured for 120 h, and  $\mu^{\text{app}}$  was determined at t = 24 to 72 h, and t = 72 to 120 h of culture. Decreased  $\mu^{app}$ at t = 72 to 120 h was observed in comparison with t = 24 to 72 h for the whole range of  $N_{72}$  and  $N_{24}$ , respectively, indicating growth inhibition during latestage of culture (Fig. 1.3B, C). The highest  $\mu^{app}$ observed the was in aggregates containing  $(1.0 \pm 0.2) \times 10^2$  to  $(1.4 \pm 0.5) \times 10^3$  cells/aggregate during late-stage of culture. To investigate the effect of aggregate size on spatial distribution of proliferating hiPSCs inside the aggregates, histological analysis of total nuclei, and cell proliferation-



Fig. 1.3 Growth profiles of single hiPSC aggregate. (A) Expansion ratio of hiPSC after 120 h culture. (B) Growth rate of hiPSC aggregates containing different initial cell numbers from t = 24 to 72 h, and (C) from t = 72 to 120 h. Data presented as practical number of cells/aggregate.

associated marker, ki-67, was performed with different cell numbers ( $N_{48}$ ) at t = 48 h. Small size aggregates containing  $N_{48} = (2.0 \pm 0.2) \times 10^1$  cells showed weak expression of ki-67 (**Fig. 1.5A**– **C**), whereas medium size aggregates containing  $N_{48} = (2.8 \pm 0.2) \times 10^2$  cells showed homogeneous expression of ki-67 across the whole aggregate regions (**Fig. 1.5D–F**). Conversely, large size aggregates containing  $N_{48} = (1.3 \pm 0.2) \times 10^3$  cells showed lower expression of ki-67 at the center region in comparison with the periphery (**Fig. 1.5G–I**).



**Fig. 1.4** Effect of conditioned medium on small size aggregate growth. Growth rate of hiPSC aggregates containing  $(2.8 \pm 0.5) \times 10^1$  and  $(8.8 \pm 0.7) \times 10^1$  cells/aggregate after culturing for 72 h with fresh medium (open bar) and conditioned medium (close bar). \* p < 0.05.

Expression of collagen type I, an ECM member found in hiPSC aggregates, was determined in specimens collected at t = 24, 72, and 120 h for medium size aggregates containing (6.8 ± 1.1) × 10<sup>2</sup> cells/aggregate (**Fig. 1.6**). Collagen type I was distributed homogeneously throughout the aggregate at t = 24, and 72 h, with accumulation at the peripheral region at t = 120 h, indicating formation of a shell-like structure during late-stage of culture. Moreover, the frequency of DAPI-positive hiPSCs in the center region of the aggregate at t = 120 h was lower than that at t = 24, and 72 h, suggesting reduced growth over time at the center region. To assess ECM accumulation, similar size aggregates at  $N_{24} = (1.8 \pm 0.4) \times 10^3$  and  $N_{120} = (1.4 \pm 0.4) \times 10^3$ 



**Fig. 1.5** Florescent images of hiPSC aggregates after 48 h of culture, showing for nuclei (red) and ki-67 (green). (A-C) aggregates containing  $(2.0 \pm 0.2) \times 10^1$  cells, (D-F) aggregates containing  $(2.8 \pm 0.2) \times 10^2$  cells, and (G-I) aggregates containing  $(1.3 \pm 0.2) \times 10^3$  cells. Scale bars: 100 µm.

cells/aggregate were investigated for collagen type I expression after culturing for 24, and 120 h, respectively (**Fig. 1.7**). A higher amount of collagen type I accumulated in the peripheral region of  $N_{120} = (1.4 \pm 0.4) \times 10^3$  cells/aggregate in comparison to  $N_{24} = (1.8 \pm 0.4) \times 10^3$  cells/aggregate.



**Fig. 1.6** Florescent images of hiPSC aggregate containing  $(6.8 \pm 1.1) \times 10^2$  cells for nuclei (red), and collagen type I (green). (A-C) aggregates cultured for 24 h, (D-F) aggregates cultured for 72 h, and (G-I) aggregates cultured for 120 h. Panels C, F, and I show magnified views of the boxed areas. Scale bars: 100 µm.

Furthermore, hiPSC aggregates containing  $N_{120} = (1.4 \pm 0.4) \times 10^3$  cells/aggregate showed lower expression of ki-67 in the center region in comparison to  $N_{24} = (1.8 \pm 0.4) \times 10^3$ cells/aggregate (**Fig. 1.8**). To confirm the undifferentiated state of hiPSCs, flow cytometry was performed, and similar frequency of Octamer-binding transcription factor (OCT) 3/4 was observed during t = 24 h (98.3% ± 1.8%), 72 h (99.6% ± 0.3%), and 120 h (99.7% ± 0.3%) of culture.



**Fig. 1.7** Florescent images of hiPSC aggregates for nuclei (red), and collagen type I (green). (A-C) show  $(1.8 \pm 0.4) \times 10^3$  cells/aggregate cultured for 24 h, and (D-F) show  $(1.4 \pm 0.4) \times 10^3$  cells/aggregate cultured for 120 h. Panels C and F are the magnified views of boxed areas. Scale bars: 100 µm.



**Fig. 1.8** Florescent images of hiPSC aggregates for nuclei (red), and ki-67 (green). (A-C) show  $(1.8 \pm 0.4) \times 10^3$  cells/aggregate cultured for 24 h, and (D-F) show  $(1.4 \pm 0.4) \times 10^3$  cells/aggregate cultured for 120 h. Scale bars: 100 µm.

#### **1.4 Discussion**

Elucidation of the aggregate growth properties is important for hiPSC expansion in suspension culture. This study quantitatively evaluated the growth properties of single hiPSC aggregates with respect to initial cell number and culture time. Factors that reduce the growth rate of hiPSC aggregates during early- and late-stage of culture were also investigated (**Fig. 1.9**).

Aggregate size played an important role because hiPSC growth rate decreased in both small and large size aggregates during t = 24 to 72 h of culture (Fig. 1.3B). In the small size aggregates, lower  $\alpha$  was observed in comparison to medium size aggregates (Fig. 1.2A). E-cadherin is coexpressed with other undifferentiation markers and plays a central role in cell survival (Li et al., 2012). Ohgushi et al. reported high cell loss after dissociating hESCs into single cells due to disruption of cell-cell connections mediated by E-cadherin (Ohgushi et al., 2010). In addition, Xu et al. suggested that E-cadherin played a central role in the survival and self-renewal of hESCs since cell death occurred due to irreparable disruption of E-cadherin following enzymatic dissociation (Xu et al., 2010). Together with these suggestions, low cell-cell connection mediated by E-cadherin in small size aggregates was considered to lead low level of cell survival rate. Moreover, small size aggregates showed a lower  $\mu^{app}$  in comparison with the medium size aggregates at t = 24 to 72 h (Fig. 1.3B). Because small size aggregates contained fewer cells in comparison with the medium size aggregates, it was assumed that small size aggregates had low secretion of growth-promoting factors in comparison to medium size aggregates. When small size aggregates were cultured with CM, a similar  $\mu^{app}$  was observed in comparison with the medium size aggregates (Fig. 1.4).



**Fig. 1.9** Schematic illustration of the present study focusing on the growth of a single hiPSC aggregate considering the effect of (**A**) aggregate size and (**B**) culture time.

CM is known to contain a high concentration of autocrine factors. In fact, hPSCs secrete many essential components via cell–cell interactions especially basic fibroblast growth factor (bFGF), TGF- $\beta$ 1, and insulin as autocrine factors which act to maintain culture homeostasis, and thus stimulate cell growth (Bendall *et al.*, 2007; Greber *et al.*, 2007; Montes *et al.*, 2009). It is

therefore hypothesized that the CM containing autocrine factors was responsible for the enhanced growth of the small size aggregates.

In the large size aggregates,  $\mu^{app}$  was decreased with increased  $N_{24}$  at t = 24 to 72 h (Fig. 1.3B), and expression of ki-67 was suppressed in the center region (Fig. 1.5). It was assumed that diffusion of oxygen and nutrients from the periphery to the center were limited, resulting in reduced cell growth and enhanced cell necrosis in the center region (Novosel et al., 2011; Jain et al., 2005). In the present study, the maximum D of large size aggregates was more than 400 µm (Fig. 1.2B) which was assumed to cause low diffusive transport of oxygen and nutrients into the densely agglomerated cells in the center region of aggregates. It has already been reported that oxygen concentration in the center region of larger embryoid bodies (EBs) (400 µm) is 50% lower in comparison to that in medium size EBs (200 µm) (Kinney et al., 2011; Winkle et al., 2012; Wu et al., 2014) which triggers apoptosis of the core cells due to low oxygen diffusion (Carmeliet et al., 1998). Moreover, the morphology and cell-cell connections are reported to alter at high cell number, which in turn stimulates the Hippo pathway signaling, triggering the phosphorylation of Yap in the nucleus, and reducing cell proliferation (Ota et al., 2008; Wada et al., 2011). Together with this suggestion, low expression of ki-67 in the center region of large size aggregate was considered to occur due to cell contact inhibition of proliferation because the cells entered to a quiescent state during early-stage of culture (Fig. 1.9A). However, the higher expansion ratios observed in the medium size aggregates containing  $N_{24} = (8.8 \pm 0.8) \times 10^1$  to  $(2.7 \pm 0.2) \times 10^2$  cells after culturing for 120 h, indicated the feasible range for initial cell seeding number for higher proliferation of hiPSC in aggregate culture (Fig. 1.3A).

Culture time was also considered as an important factor for hiPSC growth because  $\mu^{app}$  at t =72 to 120 h was lower than that at t = 24 to 72 h for the whole range of  $N_{72}$  and  $N_{24}$ , respectively (Fig. 1.3C). The highest  $\mu^{app}$  in the aggregates containing  $(1.0 \pm 0.2) \times 10^2$  to  $(1.4 \pm 0.5) \times 10^3$ cells indicated the time-dependent boundary for high proliferation of cells during late-stage of culture. Immunofluorescent staining revealed that a higher amount of collagen type I was accumulated at the aggregate periphery at t = 120 h in comparison with t = 24 and 72 h of culture (Fig. 1.6). Collagen precursors are predominantly biosynthesized in the rough endoplasmic reticulum of the cell, and secreted via a transmembrane protein transport system to the extracellular space where they accumulate, and form a network with other ECM components (Ishikawa et al., 2016). However, accumulation of collagen type I was time-dependent as a higher amount of collagen type I was observed at the peripheral region of the medium size aggregates in comparison with the large size aggregates after 120 and 24 h culture, respectively (Fig. 1.7). Due to the formation of a collagen-rich, shell-like structure on the periphery of hiPSCs aggregate, the diffusive transport of soluble factors, especially cytokines including bFGF, TGF- $\beta$ 1, as well as the diffusion of oxygen and toxic metabolites to and from the aggregate was suggested to hinder during late-stage of culture (Fig. 1.9B). hESC aggregates were also reported to form a collagen-rich, shell-like structure on their surface hindering the diffusive transport of inductive biochemicals during late-stage of culture (Sachlos et al., 2008). Moreover, reduction in the diffusive transport of nutrients was suggested to lead to the activation of apoptosis-inducing factors of the core cells (Joza et al., 2001), which likely resulted in low expression of ki-67 in the center region of the hiPSC aggregates during late-stage of culture in the present study (Fig. 1.8). Shortening the culture time ( $\sim$ 72 h) to avoid ECM shell formation on the aggregate periphery is therefore important for obtaining high growth rate during hiPSC culture. Thus, we propose that

medium size (100  $\mu$ m $-200 \mu$ m) aggregates should be maintained for better supply of nutrients and oxygen, and that the culture time should be shortened with frequent subculture to reduce ECM accumulation at the aggregates periphery for obtaining high number of hiPSCs in conventional suspension culture.

In conclusion, hiPSCs showed heterogeneous growth properties depending on aggregate size and culture time in the culture of single aggregate. Growth of small size aggregates was autocrine factor-dependent, whereas large size aggregates had low center cell proliferations due to cellular quiescence. Culture time was also important for influencing heterogeneous growth during prolonged culture period by uneven distribution of factors including oxygen, nutrients and toxic metabolites both inside and outside of the hiPSC aggregates, due to formation of an ECMshell at the periphery. Therefore, understanding the growth properties of hiPSC aggregates to maintain balance between secretion of autocrine factors and accumulation of ECM on the aggregate periphery is essential to ensure optimal growth in suspension culture. Furthermore, consideration of the boundary conditions for aggregate size and culture time is needed to prevent aggregate heterogeneity, and to obtain high growth rate during large-scale expansion of hiPSCs in suspension culture.

#### **1.5 Summary**

This study quantitatively evaluated the effects of initial cell number and culture time on the growth of hiPSCs in the single aggregate culture. Small size aggregates showed a lower growth rate in comparison to medium size aggregates during early-stage of culture (24–72 h). However, when small size aggregates were cultured in conditioned medium, their growth rate increased significantly. On the other hand, large size aggregates showed a lower growth rate and lower expression level of proliferation marker (ki-67) in the center region of aggregate in comparison to medium size aggregate during early-stage of culture. Medium size aggregates showed the highest growth rate during early-stage of culture. Furthermore, hiPSCs proliferation was dependent on culture time because the growth rate decreased significantly during late-stage of culture (72–120 h) at which point collagen type I accumulated on the periphery of aggregate, suggesting blockage of diffusive transport of nutrients, oxygen and metabolites into and out of the aggregates. Consideration of initial cell number and culture time are important to maintain balance between autocrine factors secretion and extracellular matrix accumulation on the aggregate periphery to achieve optimal growth of hiPSCs in single aggregate culture.

#### **Chapter 2**

## Botulinum hemagglutinin-mediated *in situ* break-up of human iPS cell aggregates for high-density suspension culture

#### **2.1 Introduction**

hPSCs comprising of hESCs and hiPSCs are potential source of specialized cells for cell therapy application owing to their self-renewal (Takahashi *et al.*, 2007) and differentiation capability to any types of mature cells of human body (Chambers *et al.*, 2009; Zhang *et al.*, 2009). hiPSCs are superior to hESCs for clinical applications because hiPSCs circumvents the complication of ethical issues and criticism associated with their use in research. To realize their potential, bio-process developments for the robust expansion of hiPSCs are important, since high numbers of hPSCs-derived targeted cells are required for cell therapy application (Lock and Tzanakakis, 2007; Jing *et al.*, 2008). The large-scale expansion of hiPSCs in adherent culture is difficult; therefore, suspension culture is garnering increasing attention because high cell density and better control of process parameters can be obtained in stirred suspension bioreactor culture (Olmer *et al.*, 2012; Haraguchi *et al.*, 2015).

Since large size of hiPSC aggregates show low growth rate, aggregates are dissociated into single cells for maintaining high growth rate in conventional suspension culture by the following basic steps: enzyme treatment and pipetting to dissociate the aggregates, centrifugation and washing for enzyme removal, and re-seeding into the culture vessel (Olmer *et al.*, 2010; Haraguchi *et al.*, 2015). After seeding single cells in suspension culture, cells encounter the neighboring cells and establish cell–cell connections, mediated by E–cadherin, and form spherical aggregates (Sugawara *et al.*, 2010; Li *et al.*, 2012). However, some cells fail to
encounter the neighboring cells and undergo apoptosis (Watanabe *et al.*, 2007; Ohgushi *et al.*, 2010). As a result, massive cell loss occurs during hiPSC aggregate dissociation with enzymatic digestion (Singh *et al.*, 2010; Haraguchi *et al.*, 2015). Moreover, aggregate size increases with the increase in culture time (Bauwens *et al.*, 2008; Otsuji *et al.*, 2014), and large size aggregates show undesirable differentiation and cell necrosis within the aggregate, as collagen–rich shell forms on the outer surface of the aggregate, hindering the diffusive transport of oxygen and nutrients (Sachlos *et al.*, 2008; Kinney *et al.*, 2011; Winkle *et al.*, 2012; Wu *et al.*, 2014). In the previous chapter, we have also observed decrease of hiPSC proliferation as a result of aggregate size limitation, and accumulation of collagen type I on the aggregate periphery reducing the growth rate of hiPSCs during late–stage of culture. These heterogeneous phenomena of aggregate growth lead to low cell density in conventional suspension culture (Amit *et al.*, 2010; Abbasalizadeh *et al.*, 2012; Olmer *et al.*, 2012). Therefore, routine break–up of aggregates into smaller ones is important for maintaining high growth rates to obtain a high–density of hiPSCs during long–term suspension culture.

In this study, we have developed a simple and scalable method for the expansion of hiPSCs by aggregate break–up into small ones using botulinum hemagglutinin (HA), and have illustrated the mechanism of aggregate break–up in suspension culture. This method allowed high–density culture of hiPSCs without the need for enzymatic treatment or centrifugation by simplifying culture operations in suspension.

### **2.2 Materials and Methods**

### 2.2.1 Cells and Culture Conditions

The hiPSC line, Tic, provided by the Japanese Collection of Research Bioresources (JCRB1331, JCRB Cell Bank, Japan), was routinely maintained on polystyrene substrate coated with recombinant laminin-511 E8 fragments (iMatrix<sup>TM</sup>-511; Nippi Inc., Japan) in commercially available medium (mTeSR<sup>TM</sup>1; STEMCELL Technologies, Canada). For subculture, single cells were seeded with 10  $\mu$ M ROCK inhibitor (Y-27632; Wako Pure Chemical Industries, Japan). The initial seeding was fixed at a viable cell density of  $1.0 \times 10^4$  cells cm<sup>-2</sup>. Cells were incubated at 37°C in a humidified atmosphere with 5% CO<sub>2</sub>, and the medium was exchanged daily with fresh medium. On day 4, when cells reached confluence, cells were subcultured. Briefly, hiPSCs were treated with 5 mM EDTA/PBS with 10  $\mu$ M ROCK inhibitor for 7 min at room temperature. Dissociation reagent (TrypLE Select<sup>TM</sup>, Invitrogen, USA) with 10  $\mu$ M ROCK inhibitor was then added for another 7 min at room temperature, and hiPSC colonies were dissociated into single cells by pipetting. After centrifugation, the supernatant was discarded, and cells were resuspended in fresh medium. Viable cells were counted with a cell counter (TC20; Bio-Rad Inc., USA) by the trypan blue exclusion method, and cells were re-seeded in a new culture dish.

### 2.2.2 HA Addition Conditions for Aggregate Break-up

HA complexes (His-HA, His-BHA1-FLAG: His-BHA2: Strep-BHA3) were kindly provided by Professor Yukako Fujinaga (Kanazawa University, Japan). The procedure used for reconstituting the functional HA complex was similar to that described elsewhere (Sugawara *et al.*, 2014). To determine HA addition time, hiPSC aggregates were cultured in 30 mL bioreactor (BWV-S03A; Able Co., Japan) at an initial seeding density of  $1.0 \times 10^5$  cells mL<sup>-1</sup> with 10  $\mu$ M ROCK inhibitor at an agitation rate of 55 rpm, and incubated at 37° C with 5% CO<sub>2</sub> (**Fig. 2.1**). At 96 h, hiPSC aggregates  $(5.0 \times 10^5 \text{ cells mL}^{-1})$  were dispensed in a 24-well plate (Corning Inc., USA) containing 10, 20, and 30 nM HA in the culture medium, and observed at 0, 6, 9, 12, and 24 h. In each condition, aggregates were broken up by pipetting, and images were captured using a confocal microscope (Olympus, Japan) before and after pipetting. To investigate the mechanisms of aggregate break-up by HA, aggregates were immunostained for E-cadherin and HA at each sampling point.



**Fig. 2.1** Schematic diagram showing the experimental procedures for aggregate break-up with HA.

### 2.2.3 Suspension Cultures of hiPSCs

To investigate the effect HA on hiPSC aggregate break-up, cells were detached from the culture substrate by following the method described before. After dissociating the hiPSC colony into single cells,  $1.0 \times 10^5$  cells mL<sup>-1</sup> were seeded with 10  $\mu$ M ROCK inhibitor in 30 mL bioreactor. At 87 h, HA (20 nM) was added to the culture medium and kept for 9 h (**Fig. 2.2**). Then, aggregates were broken into small sizes by pipetting, and re-seeded in 30 mL bioreactor at a seeding density of  $1.0 \times 10^5$  cells mL<sup>-1</sup> with 10  $\mu$ M ROCK inhibitor, with or without washing to remove HA from the medium, and cultured for an additional 96 h. In the control culture,

aggregates were treated with dissociation reagent (ACCUMAX; STEMCELL Technologies, Canada) with 10 µM ROCK inhibitor for 10 min. Single cells were re-seeded in 30 mL bioreactor at  $1.0 \times 10^5$  cells mL<sup>-1</sup> with 10  $\mu$ M ROCK inhibitor, and cultured for an additional 96 h. In both cases, medium was exchanged with fresh one at every 24 h.

For high-density culture of hiPSC aggregates, single cells with 10 µM ROCK inhibitor were seeded in 30 mL bioreactor  $(1.0 \times 10^5 \text{ cells mL}^{-1})$  to prepare aggregates. At 87 h, HA (20 nM) was added to the culture medium and kept for 9 h. Next, aggregates were broken into small sizes by pipetting, and all the cells were re-seeded in 30 mL bioreactor with 10 µM ROCK inhibitor and cultured for an additional 96 h. In the control culture, aggregates were not treated with HA, or not broken into small sizes, and cultured for an additional 96 h. The culture medium was exchanged with fresh medium at frequent intervals to reduce the toxic effect of metabolites during late-stage of culture.



Fig. 2.2 Schematic diagram showing the experimental procedures for hiPSCs aggregate passage in suspension culture.

### 2.2.4 Kinetic Analysis of Growth Properties of hiPSC Aggregates

To determine the apparent specific growth rate,  $\mu^{app}$  (h<sup>-1</sup>), hiPSC aggregates were collected from the bioreactor (3 samples each time) at every 24 h, and dissociated into single cells by using TrypLE Select. After centrifugation, the supernatant was discarded and cells were re-suspended in fresh medium. Live cells were counted using the cell counter. The initial live cell ratio,  $\alpha$  (-), was determined by  $\alpha = X_{24}/X_0$ , where  $X_0$  is the cell density at seeding, and  $X_{24}$  is the cell density at t = 24 h. The  $\mu^{app}$  was calculated by the following equation:  $\mu^{app} = \ln(X_f/X_i)/(t_f - t_i)$ , where  $X_f$  is the final cell density at  $t_f = 87$  or 192 h, and  $X_i$  is the initial cell density at  $t_i = 24$  or 120 h. Expansion fold was calculated by  $X_f/X_i$ , where  $X_f$  is the final cell density at  $t_f = 87$  or 192 h, and  $X_i$  is the initial cell density at  $t_i = 0$  or 96 h. The yield on culture medium, Y (cells mL<sup>-1</sup>), was calculated by the following equation:  $Y = N_i/V_m$ , where  $N_t$  is the total number of cells (cells), and  $V_m$  is the total volume of medium consumption (mL). To determine the aggregate size, images were captured every 24 h using an image analyzer with a 4× objective lens (IN Cell Analyzer 2000; GE Healthcare, USA), and aggregate diameter was measured by using Image-Pro Plus software (Version 6.0; Media Cybernetics, USA).

### 2.2.5 Analysis of Medium Components

To analyze the components of culture medium associated with cell cultivation, medium was collected at every time during medium change, and analyzed for glucose and lactic acid by using a biochemical analyzer (Bioprofile 400; Nova Biomedical, USA).

### 2.2.6 Flow Cytometry

To perform flow cytometry, cells were collected at t = 192 h for checking the expression of the pluripotency markers OCT 3/4 and SSEA 4. Briefly, single cells were prepared, treated with the Cytofix/Cytoperm<sup>TM</sup> permeabilization kit (BD Biosciences, USA), and incubated with primary antibody (mouse monoclonal OCT 3/4, Santa Cruz Biotechnology, USA), and secondary antibody (Allophycocyanin (APC) anti-mouse IgG, BD Biosciences, USA). Direct staining of stage-specific embryonic antigen (SSEA) 4 was performed by using human/mouse SSEA 4 fluorescein (R&D Systems, USA). Flow cytometry was performed with a flow cytometer (Sysmex, USA), and data were analyzed using commercially available software (FlowLogic, Inivai Technologies, Australia).

#### 2.2.7 Fluorescence Staining

The procedure for staining was similar to that described previously (Kim *et al.*, 2014). Briefly, hiPSC aggregates were transferred to a 24-well plate containing HA in the culture medium and incubated for 0, 6, 9, 12, and 24 h (**Fig. 2.1**). Then embedded in optimal cutting temperature compound (Tissue-Tek; Sakura Fine Chemical Co., Ltd., Japan), and sectioned into 20–µm–thick slices using a cryostat microtome (Leica, Germany). The sections were washed with PBS and fixed with 4% paraformaldehyde (Wako Pure Chemical Industries) for 10 min at room temperature. The specimens were washed with PBS, and permeabilized with PBS containing 0.5% Triton X-100 (Wako Pure Chemical Industries) for 5 min; washed twice with PBS containing 0.1% Tween 20 (Sigma-Aldrich); and blocked in Block Ace for 90 min at room temperature. The sections were washed with Tris-buffered saline (TBS; Dako) and then probed with anti-E-cadherin antibody (24E10; Cell Signaling Technology, USA) and anti-Flag M2

antibody (Sigma-Aldrich) to detect HA, and incubated overnight at 4°C. Next, they were washed twice with TBS and immuno-labeled with Alexa Fluor 488-conjugated goat anti-rabbit or Alexa Fluor 594-conjugated goat anti-mouse IgG for 60 min at room temperature. The cell nuclei were stained with 4', 6-diamidino-2-phenylindole (DAPI; Life Technologies) for 20 min at room temperature. Images were captured using a confocal laser scanning microscope (FV 1000; Olympus) with a 60× objective lens.

### 2.2.8 Statistical Analysis

Data presented in this study were obtained from three independent experiments and expressed as mean  $\pm$  standard deviation (SD). Statistical comparisons were evaluated using the Student's t-test and values of p < 0.01 and p < 0.05 were considered to be significant.

### **2.3 Results**

### 2.3.1 hiPSC Aggregates Break-up into Small Sizes by HA

To break-up hiPSC aggregates into smaller ones, the aggregates were cultured in medium containing 10, 20, and 30 nM HA for 0, 6, 9, 12, and 24 h (**Fig. 2.3**). At 10 nM HA, hiPSCs aggregates could not be broken into small sizes, whereas, at 30 nM HA, aggregates were broken into small sizes earlier than at 20 nM HA, and the activity of HA persisted longer. At 20 nM HA, aggregates could be broken into small sizes only for a short period (at 9h). Moreover, hiPSC aggregates showed time-dependent break-up after HA addition in suspension culture. The aggregates could not be broken up shortly after HA addition (0–6 h) or even, long after it (12–24 h). The aggregates were easily broken up into smaller ones by pipetting 9 h after HA addition.



**Fig. 2.3** Fluorescent images of hiPSC aggregates for determining HA concentration, and timing for breaking up the aggregates into small sizes by pipetting in suspension culture. Scale bar: 200 μm.



**Fig. 2.4** Immunostaining of hiPSC aggregates for E-cadherin (red) and HA (green) in suspension culture. Panels (A-J) are magnified views of merge images of aggregate centre and peripheral cells at (A, B) just after addition, (C, D) 6 h, (E, F) 9 h, (G, H) 12 h, and (I, J) 24 h of HA addition. Arrow heads indicate the localization of HA in the cytoplasm and intercellular barrier of cells. Scale bars: 50 µm.

To investigate HA localization and E-cadherin expression in the intercellular barrier, hiPSC aggregates were stained with anti-Flag M2 antibody for visualizing FLAG-tagged HA1 and HA2, and anti-E-cadherin antibody, respectively (Fig. 2.4). HA localization was observed in the peripheral cells of the aggregate, and E-cadherin expression was observed throughout the aggregate immediately upon adding HA to the culture medium (Fig. 2.4A and B). Six hours after HA addition, a higher localization of HA was observed in the peripheral cells than in the center cells of the aggregate, whereas lower expression of E-cadherin was observed in the peripheral cells compared to the center cells (Fig. 2.4C and D). However, HA localization was observed only in the intercellular barrier of the center cells of the aggregate (arrow head) 6 h after HA addition, indicating the specific binding of HA to E-cadherin. Nine hours after HA addition, low E-cadherin expression and high HA localization were observed throughout the aggregate (Fig. 2.4E and F). Furthermore, 12 h after HA addition, E-cadherin expression increased and HA localization decreased in the intercellular barrier of center cells compared to the peripheral cells of aggregate (Fig. 2.4G and H). HA localization was observed in the cell cytoplasm (arrow head) 12 h after HA addition. Twenty four hours after HA addition, high expression of E-cadherin and low localization of HA were observed throughout the aggregate (Fig. 2.4I and J).

# 2.3.2 Effect of aggregate break-up by HA on hiPSCs growth

To investigate the effect of aggregate breakup by HA, hiPSCs aggregates were prepared by seeding single hiPSCs in 30 mL bioreactor. At 96 h, aggregates were broken with HA or enzymatic digestion. hiPSCs showed a higher live cell ratio, as well as higher cell density with both HA addition (1.3  $\pm$  0.3, and (1.3  $\pm$  0.2)  $\times$  10<sup>6</sup> cells mL<sup>-1</sup>, respectively), and HA addition and removal  $(1.3 \pm 0.2, \text{ and } (1.3 \pm 0.3) \times 10^6 \text{ cells mL}^{-1},$ respectively) than with enzymatic digestion (0.7  $\pm$  0.1, and (5.5  $\pm$  1.3)  $\times$  10<sup>5</sup> cells mL<sup>-1</sup>, respectively) in suspension culture (Fig. 2.5A). The size of aggregates increased with the increase of culture time and reached to an average of  $253.4 \pm 65.3 \ \mu m$  before passaging at 87 h (Fig. **2.5B**). The size of aggregates after break–up with HA was on an average  $88.2 \pm 25.3 \ \mu\text{m}$  at 120 h, and it increased with the increase in culture time with both HA treatment and enzymatic digestion.



**Fig. 2.5** Effect of hiPSC aggregates break-up with HA and enzymatic digestion. (**A**) Growth profile and (**B**) size of hiPSC aggregate broken up with HA addition (triangle), HA addition and removal (circle), and enzymatic digestion (square). (**C**) Flow cytometry analysis for OCT 3/4 (open bar) and SSEA 4 (close bar). \*p < 0.05.

	Aggregate dissociation	Aggregate break-up with	
	with enzyme	HA	
Live cell ratio, $\alpha$ (-)	$0.7 \pm 0.1$	$1.3 \pm 0.3*$	
Apparent specific growth	$3.1 \pm 0.2$	$3.2 \pm 0.1$	
rate, $\mu^{app}$ (10 <sup>-2</sup> h <sup>-1</sup> )			
Final cell density,	$5.5 \pm 1.3$	$13.2 \pm 2.3^{**}$	
$X_{\rm f} \ (10^5 \ {\rm cells \ mL}^{-1})$			
Expansion fold (-)	$5.5 \pm 1.3$	$13.2 \pm 2.3 **$	
* $p < 0.05$ , ** $p < 0.01$			

**Table 2.1.** Comparison of growth properties of hiPSC aggregate dissociated with enzyme and aggregate broken up with HA in suspension culture (at t = 96-192 h).

The  $\mu^{app}$  was similar in both HA (( $3.2 \pm 0.1$ ) ×  $10^{-2}$  h<sup>-1</sup>) and enzymatic digestion (( $3.1 \pm 0.2$ ) ×  $10^{-2}$  h<sup>-1</sup>) (**Table 2.1**). However, higher expansion fold was obtained in the culture with aggregate break-up with HA ( $13.2 \pm 2.3$ ) than with enzymatic digestion ( $5.5 \pm 1.3$ ). hiPSCs showed similar expression of OCT 3/4 (98.9% ± 0.2%, 99.1% ± 0.8%, and 98.2% ± 0.6%), and SSEA 4 (98.0% ± 2.1%, 99.2% ± 1.2%, and 97.6% ± 1.7%) in HA addition, HA addition and removal, and enzymatic digestion, respectively (**Fig. 2.5C**).

### 2.3.3 High-density Culture of hiPSC by Aggregate Break-up

To obtain high cell density in a single vessel, hiPSCs were cultured for 192 h with or without aggregate break-up by HA in suspension culture. A higher cell density of  $(4.5 \pm 0.2) \times 10^6$  cells mL<sup>-1</sup> was obtained with aggregate break-up by HA than that without aggregate break-up ( $(1.7 \pm 0.1) \times 10^6$  cells mL<sup>-1</sup>) at 192 h (**Fig. 2.6A**). Aggregate size increased gradually and reached an average of 216.9 ± 47.0 µm before breaking up at 87 h (**Fig. 2.6B**). When aggregates were broken up with HA, small size of aggregates (98.9 ± 31.7 µm) was observed at 120 h, and



**Fig. 2.6** High-density culture of hiPSC with aggregate break-up. (**A**) Growth profile, and (**B**) size of hiPSC aggregate with aggregate break-up by HA (open circle), and without aggregate break-up (close circle). Glucose consumption (open circle), and lactic acid production (close circle) in the (**C**) culture of aggregate break-up by HA, and (**D**) conventional culture without aggregate break-up. (**E**) Flow cytometry analysis for OCT 3/4 (open bar), and SSEA 4 (close bar) in the culture of aggregate break-up by HA, and in the conventional culture without aggregate break-up. \**p* < 0.01.

	Without aggregate		With aggregate	
	break-up		break-up	
	Early-stage	Late-stage	Early-stage	Late-stage
	(0–87 h)	(96–192 h)	(0–87 h)	(96–192 h)
Apparent specific growth rate,	$3.5 \pm 0.1$	$1.1 \pm 0.1$	$3.6 \pm 0.2$	$2.4 \pm 0.2*$
$\mu^{app} (10^{-2} h^{-1})$				
Final cell density,	$5.4\pm0.7$	$16.6\pm1.2$	$5.6\pm0.6$	$45.3\pm2.3*$
$X_{\rm f} (10^5 \text{ cells mL}^{-1})$				
Expansion fold (-)	$5.4\pm0.7$	$2.4\pm0.1$	$5.6\pm0.6$	$6.3\pm0.3*$
* <i>p</i> < 0.01				

**Table 2.2.** Comparison of growth properties of hiPSC with or without aggregate break-up by HA in suspension culture.

the size increased gradually thereafter. However, aggregate size did not increase significantly in the control culture without aggregate break–up during late-stage of culture (120–192 h). Higher consumption of glucose and production of lactic acid were observed in the suspension culture of hiPSCs with aggregate break–up by HA than in the control culture without aggregate break–up (**Fig. 2.6C and D**). hiPSCs cultured with the aggregate break–up with HA showed higher  $\mu^{app}$  ((2.4 ± 0.2) × 10<sup>-2</sup> h<sup>-1</sup>) than that of the control culture without aggregate break-up ((1.1 ± 0.1) × 10<sup>-2</sup> h<sup>-1</sup>) during late-stage of culture (**Table 2.2**). Higher fold of expansion (6.3 ± 0.3) was obtained in the culture with aggregate break–up compared to the control culture without aggregate break–up (2.4 ± 0.1). hiPSCs showed similar expression of OCT 3/4 (98.8% ± 0.1%, and 98.6% ± 0.5%) and SSEA 4 (98.1% ± 1.1%, and 98.5% ± 1.3%) in the culture with HA and the control culture without HA, respectively (**Fig. 2.6E**).

### **2.4 Discussion**

The spontaneous formation of aggregates after seeding single cells in culture vessel provides an advantage for large-scale expansion of hiPSCs in suspension culture. However, maintaining high growth rate of aggregates during long-term culture is difficult because aggregate size increases with an increase in culture time, and large size aggregates show heterogeneous growth owing to the occurrence of cell necrosis in the center region of the aggregate. Therefore, controlling aggregate size is required to obtain high number of cells in suspension culture. In this study, we established a simple method for breaking up aggregates into small sizes by using HA and investigated the mechanisms underlying the disruption of E-cadherin–mediated cell–cell connections by HA to obtain high-density of hiPSCs for simplifying the culture operations in suspension as shown in **Fig. 2.7**.

# 2.4.1 Temporal activity of HA facilitated the disruption of E-cadherin-mediated cell-cell connections in hiPSCs aggregate

After adding HA in the culture medium, compact aggregates turned into easy to mold which was further broken into small sizes by pipetting (**Fig. 2.3**). However, during short (0–6 h) and long (12–24 h) time after HA addition; aggregates could not be broken into small sizes. Aggregates could be broken into small sizes only 9 h after HA addition. Generally, cell-cell connections in hiPSC aggregates are mediated by E-cadherin (Chen *et al.*, 2010; Li *et al.*, 2012). While hiPSC aggregates were exposed to HA, cell-cell connections were loosened by directly disrupting the E-cadherin located in the intercellular barrier of hiPSCs aggregate (**Fig. 2.4**). HA was reported to disrupt the E-cadherin mediated cell-cell connections in different types of cells in adherent culture (Sugawara *et al.*, 2010; Lee *et al.*, 2014; Kim *et al.*, 2017). However, HA bound





density of hiPSCs in suspension culture.

to the E-cadherin of aggregate peripheral cells just after adding in the culture medium as shown in Fig. 2.7. After 6 h of addition, HA disrupted the E-cadherin mediated cell-cell connections of the peripheral cells and penetrated into the aggregate. The center cells of aggregate maintained high frequency of cell-cell connections mediated by E-cadherin after 6 h of addition since the time was not enough to reach the HA at the center region of aggregate. Furthermore, the localization of HA was observed only in the intercellular barrier of center cells of aggregate after 6 h of HA addition (Fig. 2.4C, arrow head), indicating the specific binding of HA to E-cadherin, as well as, primary way of HA transportation into the aggregate via paracellular route. Lee *et al* reported three ways of HA transportation into the epithelial cells in adherent culture: transcellular route, E-cadherin and carbohydrate-mediated basolateral route, and paracellular route (Lee et al., 2014). However, most of the E-cadherin in the intercellular barrier of hiPSC aggregate was disrupted after 9 h of HA addition, causing weak cell-cell connections which promoted breaking up the aggregates into small sizes by pipetting. The weak cell-cell connection was reported previously to stimulate the Src kinase activity which in turn activates the endocytic machinery for E-cadherin internalization into the cells (Balzac et al., 2005). Together with this suggestion, the least expression of E-cadherin after 9 h of HA addition was considered due to the internalization of E-cadherin into the cells. However, after 12 h of HA addition, aggregates could not be broken into small sizes by pipetting as E-cadherin re-expressed in the intercellular barrier of center cells of aggregate to promote strong cell-cell connections. The re-expression of E-cadherin in the center cells of aggregate was suggested due to the stimulation by Rap1 activation which promoted the recycling of internalized E-cadherin from the cell (Balzac et al., 2005; Kim et al., 2017). However, the activity of HA retained after 12 h of addition on the peripheral cells of aggregate since lower expression of E-cadherin was observed

in comparison to the center cells. The localization of HA in the cytoplasm after 12 h of addition indicated the secondary way of HA transportation into the aggregate via trans-cellular route (**Fig. 2.4G, arrow head**). After 24 h, the localization of HA decreased significantly and E-cadherin re-expressed throughout the aggregate suggesting the absorption of HA into hiPSC aggregate since binding of HA complex to cells facilitates the absorption into the cells (Balzac *et al.*, 2005). Thus, it was clarified that primary transportation of HA into the aggregate was mediated by paracellular route; whereas, secondary transportation was mediated by transcellular route after HA addition in suspension culture. The temporal activity of HA was also confirmed for breaking up aggregates into small sizes in suspension culture.

## 2.4.2 hiPSC aggregates break-up with HA had no effect on growth rate rather it enhanced live cell ratio and simplified culture operations in comparison to enzymatic digestion

hiPSC aggregates break-up by HA resulted in higher initial live cell ratio and higher cell density in comparison to aggregates dissociation with conventional enzymatic digestion (**Fig. 2.5**). Since aggregates were broken into small sizes, high cell-cell connections and ECM were assumed to maintain within the aggregates which provided a higher live cell ratio in comparison to aggregate dissociation with enzymatic digestion. In the enzymatic digestion, aggregates were dissociated into single cells which resulted in massive cell loss during initial seeding in suspension culture. Singh *et al.* also reported high percentage of apoptotic cells during initial seeding after dissociating hPSC aggregate into single cells with enzymatic digestion (Singh *et al.*, 2010). Generally, E-cadherin is co-expressed with other undifferentiation markers and plays a central role in cell survival and proliferation (Li *et al.*, 2012). Ohgushi *et al.* and Xu *et al.* reported high cell loss after dissociating hESCs into single cells due to E-cadherin disruption

following enzymatic dissociation (Ohgushi et al., 2010; Xu et al., 2010). Taken together, low cell-cell connections due to irreparable loss of E-cadherin was considered to lead to low live cell ratio in the aggregate dissociation with enzymatic digestion. On the other hand, washing of cells is mandatory in the conventional aggregate dissociation with enzymatic digestion to remove enzyme which along with remove ECM from cell suspension. Removal of ECM triggers apoptosis as ECM support growth and proliferation of hPSCs (Suh and Han, 2011; Chang et al., 2015). As a result, low expansion of cells (~5 fold) was obtained in the aggregate dissociation with enzymatic digestion in comparison to the aggregate break-up with HA (~13 fold) (Table 1). Haraguchi et al. and Singh et al. reported 2-fold expansion (Singh et al., 2010, Haraguchi et al., 2015), whereas, Zweigerdt et al. reported 3-6 fold expansion (Zweigerdt et al., 2011) of hPSCs during aggregate dissociation with enzymatic digestion in suspension culture. Since HA had no negative effect on hiPSC proliferation (Fig. 2.5), as well as in maintaining pluripotency (Fig. 2.5C), and inactivated spontaneously after addition (24 h); removal of HA from medium by washing was not needed which promoted simplification of culture operations in suspension. Therefore, a simple method for hiPSC aggregate break-up was established which can be applied for large-scale expansion of hiPSCs without washing or centrifugation during passage in suspension culture.

## 2.4.3 Overcoming size-dependency on aggregate growth resulted in high cell density and high growth rate in suspension culture

The apparent specific growth rate of hiPSC decreased significantly during late-stage (96–192 h) of control culture without aggregate break-up due to aggregate size limitation (**Table 2.2**). Large size of aggregates have been previously reported to show undesired differentiation as well

as cell necrosis in the center region of aggregate as collagen-rich shell forms on the outer surface of aggregate which hinders the diffusive transport of oxygen and nutrients during late-stage of culture (Sachlos et al., 2008; Kinney et al., 2011; Winkle et al., 2012; Wu et al., 2014). In the previous chapter, lower growth rate was observed both in large size aggregate as well as with the increase of culture time due to collagen type I formation on the aggregate periphery during latestage of culture. In this study, while hiPSC aggregates were broken into small sizes with HA in the late-stage of culture, high cell density ( $(4.5 \pm 0.2) \times 10^6$  cells mL<sup>-1</sup>) was obtained which was three times higher than the control culture without aggregate break-up (Fig. 2.6). Moreover, higher consumption of glucose and production of lactic acid was observed in the culture with aggregate break-up by HA in comparison to without aggregates break-up. Several recent studies have also been reported to obtain high-density of hPSCs ( $2.0 \times 10^6$  cells mL<sup>-1</sup>) and high growth rate in the conventional aggregate suspension culture (Amit et al., 2010; Abbasalizadeh et al., 2012; Olmer et al., 2012;), which are significantly lower in comparison to the current study. Although hiPSCs can be expanded to a high-density  $(3.6 \times 10^6 \text{ cells mL}^{-1})$  by continuous perfusion (Kropp et al., 2016), maintaining such density in aggregate suspension culture is still challenging due to aggregate size limitation. In the current study, we have succeeded to overcome the aggregate size limitation through the break-up of hiPSC aggregates into small sizes with HA to obtain high cell density with high expression of pluripotency markers (Fig. 2.6E) in suspension culture. Moreover, higher expansion fold (~6 folds) was obtained with the break-up of aggregates by HA which was higher than those previously reported for hPSCs expansion (2-4 folds) during long-term suspension culture (Abbasalizadeh et al., 2012; Chen et al., 2012; Olmer et al., 2012; Hunt et al., 2014). This successful performance suggests HA treatment as a promising technique to prevent initial cell loss during seeding, simplifying passage operations

without centrifugation or washing, and for obtaining high cell density with high yield on medium in suspension culture.

In conclusion, a simple and scalable method for *in situ* break-up of hiPSC aggregates was established by revealing the temporal activity of HA for disrupting E-cadherin mediated cell-cell connections by minimizing the initial cells loss during seeding in suspension culture. This method also allowed high density culture of hiPSCs without the need of enzymatic treatment or centrifugation in suspension culture. Thus, this promising technique of aggregate break-up with HA will lead to the establishment of a simple closed-system for robust expansion of hiPSCs in suspension culture.

### 2.5 Summary

In this study, a simple method for break-up of hiPSC aggregates into small sizes was established by using botulinum hemagglutinin (HA) which specifically bound with E-cadherin, and disrupted cell-cell connections in hiPSC aggregate. The primary transportation of HA into the aggregate was mediated by paracellular route, whereas, the secondary transportation of HA was mediated by transcellular route after HA addition in the aggregate suspension culture. Since HA showed temporary activity for aggregate break-up the, removal of HA from culture medium was not needed during suspension culture. hiPSC aggregates broken with HA showed higher number of live cells, higher final cell density, and higher yield on culture medium in comparison to the aggregate dissociation with enzymatic digestion in suspension culture. Moreover, a maximum density of  $(4.5 \pm 0.2) \times 10^6$  cells mL<sup>-1</sup> was obtained by breaking up the aggregate break-up. Therefore, the temporal activity of HA for disrupting E-cadherin mediated cell-cell connections paved the way to establish a *in situ* simple method of hiPSC aggregate break-up in bioreactor without initial cell loss which led to a high cell density in suspension culture.

### Chapter 3

### Culture medium refinement by dialysis for the expansion of human induced pluripotent stem cells in suspension

### **3.1 Introduction**

Human induced pluripotent stem cells (hiPSCs) (Takahashi *et al.*, 2007) hold great promise in the field of regenerative medicine for cell-based therapies (Schulz *et al.*, 2012), tissue engineering (Badylak *et al.*, 2011), and drug discovery because of their pluripotency and selfrenewal capacity (Wernig *et al.*, 2007). To realize their potential, the developments of bioprocesses for the expansion of hiPSCs are important, as a large number of cells are required for clinical application (Lock and Tzanakakis, 2007; Jing *et al.*, 2008; Kehoe *et al.*, 2010; Serra *et al.*, 2010; Olmer *et al.*, 2010). Compared to adherent culture, suspension culture of hiPSC aggregate is promising to obtain such numbers of cells, as a higher cell density and better control of process parameters can be achieved in a stirred-suspension bioreactor (Zweigerdt *et al.*, 2011; Olmer *et al.*, 2012; Haraguchi *et al.*, 2015).

Production of lactic acid is higher in hPSC culture than it is in other cell types (Varum *et al.*, 2011), and lactic acid has potential toxic effects on the proliferation of hPSC (Ouyang *et al.*, 2007; Chen *et al.*, 2010). As a result, frequent changing of culture medium is required to remove toxic metabolites (Fernandes-Platzgummer *et al.*, 2014), although the consumption of glucose is not high in hPSC culture compared to that in other types of cells (Terstegge *et al.*, 2007). Moreover, suspension culture of hiPSCs in a bioreactor requires a large volume of culture medium, since spent medium is replaced daily with fresh medium to reduce the accumulation of lactic acid, ammonium ( $NH_4^+$ ), and other toxic metabolites, which increase with an increase in

cell density (Abbasalizadeh *et al.*, 2012). These toxic metabolites reduce the growth of hiPSCs in suspension culture. On the other hand, hPSCs secrete various autocrine factors in the culture medium that maintain culture homeostasis and potentiate the growth of cells. hESCs have been reported to spontaneously secrete TGF- $\beta$ 1 and IGF-II in response to bFGF in feeder-free adherent cultures (Bendall *et al.*, 2007; Montes *et al.*, 2009). These autocrine factors are removed from culture medium along with toxic metabolites during daily medium changes in conventional suspension culture. The cost of culturing hiPSC increases with increasing medium consumption, as the culture medium contains many costly macromolecules including bFGF, TGF- $\beta$ 1, and insulin (Tian *et al.*, 2004; Ludwig *et al.*, 2006).

In this study, a method to refine culture medium by dialysis was established. This method allowed the removal of toxic metabolites, as well as recycling of autocrine factors in the culture medium and reduced the use of macromolecules (TGF- $\beta$ 1, insulin, etc.) for the expansion of hiPSCs in suspension culture. The minimum inhibitory level of lactic acid for hiPSCs and lactic acid removal by dialysis were also investigated.

### **3.2 Materials and methods**

### **3.2.1 Culture of hiPSCs**

hiPSCs were obtained from Japanese Collection of Research Bioresources (Tic, JCRB number: 1331, Japan). For subculture, single cells  $(1.0 \times 10^4 \text{ cells cm}^{-2})$  were seeded with 10  $\mu$ M ROCK inhibitor (Tocris Bioscience, UK) in a 55 cm<sup>2</sup> culture dish (Corning Inc., USA) coated with laminin fragment (iMatrix<sup>TM</sup>-511, Nippi Inc., Japan), with hiPSC expansion medium, mTeSR<sup>TM</sup>1 (STEMCELL Technologies, Canada), 2 mm in depth. Cells were incubated at 37°C

in a humidified atmosphere with 5% CO<sub>2</sub>, and medium was exchanged daily with fresh one. On day 4, when cells reached 80–90% confluence, cells were subcultured. For routine passage, hiPSCs were treated with 5 mM EDTA/PBS for 7 min at room temperature with 10  $\mu$ M ROCK inhibitor. After that, dissociation reagent (TrypLE Select<sup>TM</sup>, Invitrogen, USA) with10  $\mu$ M ROCK inhibitor was applied for another 7 min at room temperature. After dissociating the hiPSC colony into single cells,  $1.0 \times 10^4$  cells cm<sup>-2</sup> were seeded with 10  $\mu$ M ROCK inhibitor into a new culture dish.

### 3.2.2 Effect of lactic acid

The effect of lactic acid on hiPSC growth was studied in two-dimensional (2D) adherent culture. Media containing various initial concentrations of lactic acid (0, 0.5, 1.0, 1.5, 2.0 and 2.5 g L<sup>-1</sup>) were used to culture hiPSCs in 6-well plates (Corning Inc., USA) coated with laminin at a seeding density of  $1.0 \times 10^4$  cells cm<sup>-2</sup>. The initial pH of the medium was adjusted with sodium hydroxide (NaOH) to 7.4. To determine the adhesion ratio,  $\alpha$  and the apparent specific growth rate,  $\mu^{app}$ , live cells were harvested, and dissociated into single cells by using TrypLE select. Then live cells were counted by using a cell counter (TC20, Bio-Rad Inc., USA) with trypan blue exclusion method. The adhesion ratio was determined by the equation,  $\alpha = X_{24}/X_0$ , where  $X_0$  and  $X_{24}$  are cell densities at seeding and t = 24 h, respectively. The apparent specific growth rate,  $\mu^{app}$  (h<sup>-1</sup>), was calculated by the following equation:  $\mu^{app} = \ln(X_f/X_i)/(t_f - t_i)$ , where  $X_f$  is the final cell density at  $t_f = 72$  h, and  $X_i$  is the initial cell density at  $t_i = 24$  h. Flow cytometry was performed to detect the frequency of pluripotency markers of hiPSC, OCT 3/4, and TRA-1-60 at t = 72 h.

### 3.2.3 Medium refinement system

The medium refinement system was designed with a cross flow dialysis membrane filter by which used suspension culture medium was refreshed with DMEM/F-12 (Sigma-Aldrich, USA) as the dialysate. The cross flow dialysis membrane filter contained 300 hollow fibers, and had a 200  $\mu$ m internal diameter, a MWCO of 10 kDa, and a surface area of 150 cm<sup>2</sup> (AKH-5S, Toyobo, Japan) (**Fig. 3.1**). Aseptic connections were maintained using silicone tubing. The used medium was collected from the aggregate suspension culture through a circular filter inside the bioreactor, which was prepared with hollow fibers (surface area, 25 cm<sup>2</sup>; internal diameter, 1.3 mm; MFC-K1, Toyobo). The flow rate of used medium was 2.6 mL min<sup>-1</sup>, whereas, the cross flow rate of fresh basal medium (DMEM/F12), which is considered to be equivalent to the basal component in expansion medium (Ludwig *et al.*, 2006), was 6.6 mL min<sup>-1</sup>. To analyze metabolites and cytokines, medium was sampled before and after dialysis. The concentrations of glucose, lactic acid and ammonium were analyzed using a biochemical analyzer (Bioprofile 400, Nova Biomedical). The concentrations of cytokines, specifically bFGF and TGF- $\beta$ 1, in the culture medium were determined by an enzyme-linked immunosorbent assay (ELISA) kit (R&D



Fig. 3.1 Schematic diagram of medium refinement system for hiPSC culture in a stirredsuspension bioreactor.58

### 3.2.4 Adherent culture of hiPSCs with refined medium

Human iPSCs were cultured in a 6-well plate with fresh medium, refined medium with or without bFGF (100 µg L<sup>-1</sup>, DS Pharma, Japan), and unrefined medium for 72 h. To determine the values of  $\alpha$  and  $\mu^{app}$ , live cells were counted at t = 24 and 72 h. Flow cytometry was used to detect OCT 3/4 and TRA-1-60 expression at t = 72 h.

### 3.2.5 Suspension culture with the medium refinement system

Suspension culture of hiPSC aggregates was performed by seeding single cells at a density of  $2.0 \times 10^5$  cells mL<sup>-1</sup> in a stirred suspension bioreactor (working volume: 100 mL, BWV-S10A, Able Co., Japan) with 10 µM ROCK inhibitor (Fig. 3.1). In the expansion medium, dissolved oxygen (DO, 3.2 mg  $L^{-1}$ ) was regulated by air and N<sub>2</sub> gas. Moreover, pH was regulated by CO<sub>2</sub> and the agitation rate was maintained at 40 rpm. In the control culture, medium was exchanged with fresh medium every 24 h during the first 48 h and every 12 h thereafter. During culture with the medium refining system, spent medium was exchanged with fresh medium only at t = 24 h to remove the ROCK inhibitor. At t = 48 h, hiPSC aggregates, along with the used medium, were transferred to another 100 mL bioreactor which was equipped with the medium refinement system as the cell separation filter inside the bioreactor interrupted the formation of aggregates during the initial seeding of single cells at t = 24 h. Further culture was performed by dialyzing the used medium with the basal medium (600 mL at each time) supplemented with bFGF (100  $\mu$ g L<sup>-1</sup>) every 12 h, and TGF- $\beta$ 1 (1  $\mu$ g L<sup>-1</sup>, Peprotech, USA) every 24 h. Heparin (100  $\mu$ g L<sup>-1</sup>, STEMCELL Technologies, Canada) was added to both the control culture and medium refinement culture to stabilize bFGF. This experiment was repeated three times separately with three independent bioreactors. To determine specific growth rate,  $\mu$  (h<sup>-1</sup>), hiPSC aggregates were

collected from bioreactor (3 samples in each time) at t = 24, 48, 72, and 96 h, and dissociated into single cells by using TrypLE select. Then live cells were counted by using a cell counter with trypan blue exclusion method. The apparent specific growth rate,  $\mu^{app}$  (h<sup>-1</sup>), was calculated by the following equation:  $\mu^{app} = \ln(X_f/X_i)/(t_f - t_i)$ , where  $X_f$  is the final cell density at  $t_f = 96$  h, and  $X_i$  is the initial cell density at  $t_i = 24$  h. Culture medium was sampled every 12 h for analyzing metabolites (glucose and lactic acid). At t = 96 h, flow cytometry was performed for checking the expression of OCT 3/4 and TRA-1-60. To determine the amounts of remaining growth factors in the culture medium, used medium was collected at t = 24 and 96 h, and analyzed by ELISA.

### **3.2.6 Flow cytometry**

Cells were treated with the Cytofix/Cytoperm<sup>™</sup> permeabilization kit (BD Biosciences), and incubated with primary antibody, mouse monoclonal OCT 3/4 (Santa Cruz Biotechnology), and secondary antibody, APC anti-mouse IgG (BD Biosciences). Direct staining of TRA-1-60 was performed using FITC-conjugated mouse anti-human TRA-1-60 (BD Pharmingen, USA). Flow cytometry was performed with Japan-made sorter, analyzer (JSAN, Bay Bioscience, Japan), and data were analyzed by commercially available software (FlowLogic, Inivai Technologies, Australia).

### **3.2.7 Stabilization of bFGF**

For bFGF stabilization, fresh medium was kept at 4°C and 37°C in a 30 mL bioreactor (BWV-S03A, Able Co., Japan) with or without heparin solution (100  $\mu$ g L<sup>-1</sup>) for 24 h, and then analyzed by ELISA (Furue *et al.*, 2008; Chen *et al.*, 2012). To investigate the consumption of

bFGF, hiPSC aggregates  $(5.0 \times 10^5 \text{ cells mL}^{-1})$  were cultured in a 30 mL bioreactor with fresh medium supplemented with heparin. After 24 h of culture, the used medium was collected and analyzed by ELISA.

### 3.2.8 Statistical analysis

Data presented in this study were obtained from three independent experiments and expressed as the mean  $\pm$  standard deviation (SD). Comparisons were evaluated statistically using Student's *t*-test, and values of *p* < 0.05 and *p* < 0.01 were considered significant.

### **3.3 Results**

### 3.3.1 Effect of lactic acid on hiPSC growth and removal of lactic acid by dialysis

To investigate the effects of lactic acid on hiPSC growth, various concentrations of lactic acid were added to the culture medium, and cells were evaluated. Lactic acid was found to have negative effects on hiPSC growth, and  $\mu^{app}$  decreased with increasing lactic acid concentrations in medium in adherent culture (**Fig. 3.2A**). More than 1 g L<sup>-1</sup> of lactic acid had toxic effects, and 1.5 g L<sup>-1</sup> of lactic acid completely inhibited the growth of hiPSCs. Lower frequency of TRA-1-60 positive cells (50.1% ± 3.0%) were also observed in culture with medium containing 1.0 g L<sup>-1</sup> of lactic acid, compared to that in the control (62.6% ± 6.0%) (**Fig. 3.2B**). As the growth rate and frequency of TRA-1-60 decreased due to the toxic effects of lactic acid, removal of lactic acid from the culture medium might be beneficial for the expansion of hiPSC.



**Fig. 3.2** Effect of lactic acid on hiPSCs growth and pluripotency. (A) Apparent specific growth rate of hiPSCs, and (B) frequency of OCT 3/4 (close bar) and TRA-1-60 (open bar) at different concentrations of lactic acid in adherent culture. \*p < 0.01.

The medium refinement system, which was set to refresh 100 mL of used suspension culture medium with 600 mL of basal medium by dialysis, was able to remove lactic acid (1.18  $\pm$  0.07 g L<sup>-1</sup>) and ammonium (0.01  $\pm$  0.00 g L<sup>-1</sup>) from the used medium, and maintain a high concentration of glucose (3.20  $\pm$  0.05 g L<sup>-1</sup>) in the refined medium (**Table 3.1**).

### **3.3.2** Cytokines in the refined medium

Since the MWCO of the dialysis filter (10 kDa) was smaller than the size of most cytokines supplemented in hiPSC culture medium, almost all of the cytokines were considered to be retained after dialysis in the refined medium. In fact, 87% of bFGF and 80% of TGF- $\beta$ 1 remained in the refined medium after dialysis (**Table 3.1**). However, the concentration of bFGF in the refined medium was lower ( $1.2 \pm 0.4 \ \mu g \ L^{-1}$ ) than in the fresh medium ( $76.8 \pm 5.1 \ \mu g \ L^{-1}$ ). Further investigation of bFGF revealed that bFGF was thermally unstable at 37°C without heparin (Chen *et al.*, 2012) (**Fig. 3.3**). However, low concentrations of bFGF ( $13.9 \pm 1.1 \ \mu g \ L^{-1}$ )



**Fig. 3.3** Different storage conditions of fresh medium for checking bFGF aggregation, stabilization with heparin and consumption by cells after 24 h at 37°C. \*p < 0.01.

were detected in the hiPSC suspension culture medium even when the medium was supplemented with heparin (**Fig. 3.3**). From this result, consumption of bFGF by hiPSC was confirmed in suspension culture.

Medium components	Fresh medium	Dialysis medium (DMEM/F-12)	Used medium	
(mTeSR <sup>IM</sup> I)	(mTeSR <sup>IM</sup> I)		Before dialysis	After dialysis
Glucose (g $L^{-1}$ )	$3.25\pm0.06$	$3.40\pm0.10$	$1.74\pm0.55$	$3.20\pm0.05*$
Lactic acid $(g L^{-1})^{\dagger}$	N. D.	N. D.	$1.18\pm0.07$	N. D.*
Ammonium (g $L^{-1}$ ) <sup>††</sup>	N. D.	N. D.	$0.01\pm0.00$	N. D.*
bFGF ( $\mu g L^{-1}$ )	$76.8\pm5.1$	N. D.	$1.32\pm0.32$	$1.16\pm0.37$
TGF- $\beta 1 (\mu g L^{-1})$	$1.40\pm0.17$	N. D.	$0.25\pm0.04$	$0.20\pm0.04$

Table 3.1 Concentration of medium components in fresh, dialysis, and used media

\*p < 0.05, N. D.: not detected, <sup>†</sup>Minimum detection limit: 0.2 g L<sup>-1</sup>, <sup>††</sup> Minimum detection limit: 0.004 g L<sup>-1</sup>

### 3.3.3 Adherent culture with refined medium

After preparing refined medium by dialysis, the medium was screened for its ability to potentiate the proliferation of hiPSC in adherent culture. hiPSC showed a higher  $\alpha$  (1.1 ± 0.1), and  $\mu^{app}$  ((0.6 ± 0.2) × 10<sup>-2</sup> h<sup>-1</sup>) in the refined medium than in the unrefined medium (**Fig. 3.4A and B**), although  $\mu^{app}$  in the refined medium was found to be lower than that in fresh medium ((3.7 ± 0.2) × 10<sup>-2</sup> h<sup>-1</sup>). When the refined medium was supplemented with bFGF, similar  $\alpha$  (1.1 ± 0.2) and  $\mu^{app}$  ((3.6 ± 0.2) × 10<sup>-2</sup> h<sup>-1</sup>) were observed as in the fresh medium. hiPSCs cultured in the refined medium supplemented with bFGF showed similar frequency of OCT 3/4 (98.6% ± 0.4%),



Fig. 3.4 Adherent culture of hiPSCs with the refined medium. (A) Adhesion ratio, (B) apparent specific growth rate, (C) Frequency of OCT 3/4 (close bar) and TRA-1-60 (open bar). \*p < 0.01, ND.: Not determined.

and TRA-1-60 (71.6%  $\pm$  4.0%) as fresh medium, whereas, refined medium without bFGF showed lower frequency of TRA-1-60 (47.4%  $\pm$  3.8%) (**Fig. 3.4C**).

### **3.3.4** Suspension culture with the medium refinement system

Suspension culture of hiPSCs was performed using both fresh medium and the medium refinement system in a stirred suspension bioreactor. In the conventional culture using fresh medium, a final cell density of  $(8.1 \pm 2.2) \times 10^5$  cells mL<sup>-1</sup> was obtained at 96 h (**Fig. 3.5A**), and the production of lactic acid and consumption of glucose were observed (**Fig. 3.5C and D**).



**Fig. 3.5** Suspension culture of hiPSC with the medium refinement system. (**A**) Growth profile of hiPSC with fresh medium (close bar), and the medium refining system (open bar). (**B**) Frequency of OCT 3/4 (close bar) and TRA-1-60 (open bar). Concentration of lactic acid (close circle), and glucose (open circle) during culture with (**C**) fresh medium and (**D**) with the medium refining system.

In the culture with the medium refinement system, a final cell density,  $X_f = (1.1 \pm 0.1) \times 10^6$  cells mL<sup>-1</sup> was obtained when the medium was refined from t = 48 h (**Fig. 3.5A**), with the production of lactic acid and consumption of glucose in suspension culture (**Fig. 3.5C and D**). In addition, hiPSC in the culture with the medium refinement system showed a similar growth rate  $((3.5 \pm 0.2) \times 10^{-2} \text{ h}^{-1})$  as that in conventional culture  $((3.1 \pm 0.2) \times 10^{-2} \text{ h}^{-1})$ . Flow cytometry analysis showed the frequency of OCT 3/4 and TRA-1-60 in conventional culture were 98.1%  $\pm$  0.9% and 69.2%  $\pm$  2.1%, respectively, whereas, the respective levels in the culture with the medium refinement system were 99.5%  $\pm$  0.2% and 78.3%  $\pm$  1.1% (**Fig. 3.5B**). When the used medium was investigated for autocrine factors, a higher concentration of bFGF was detected at t = 96 h of culture (50.1  $\pm$  0.5 µg L<sup>-1</sup>, and 64.4  $\pm$  5.6 µg L<sup>-1</sup>, in the conventional culture and culture with the medium refinement system, respectively) than at t = 24 h in the used culture medium (30.2  $\pm$  2.6 µg L<sup>-1</sup>, and 29.2  $\pm$  5.4 µg L<sup>-1</sup>, respectively) (**Fig. 3.6A**).



Fig. 3.6 Remaining amount of cytokines in used culture medium. Concentration of (A) bFGF, and (B) TGF- $\beta$ 1 during culture with fresh medium (open bar) and medium refinement system (close bar). \* p < 0.05.

Higher concentrations of TGF- $\beta$ 1 were also detected in medium after 96 h of culture (1.3 ± 0.3 µg L<sup>-1</sup>) than after 24 h (0.7 ± 0.0 µg L<sup>-1</sup>) in the culture with medium refinement system (**Fig. 3.6B**). The concentrations of bFGF and TGF- $\beta$ 1 at 96 h in the used medium were higher in the culture with medium refinement system than in the conventional culture. By applying the medium refinement system, the use of costly medium components was reduced for the expansion of hiPSCs in suspension culture (**Table 3.2**).

**Table 3.2** Theoretical amount of medium and other components required for conventional suspension culture and culture with the medium refinement system at a 100 -mL-scale

Required items	Conventional culture	Culture with the medium refinement system
Medium		
Fresh medium (mTeSR™1)	0.6 L	0.2 L
Dialysis medium (DMEM/F12)		2.4 L
Medium components		
bFGF	60 µg	60 µg
TGF-β1	1.2 µg	0.6 µg
Insulin	11.6 mg	3.9 mg
Transferrin	6.4 mg	2.1 mg
L-ascorbic acid	38.4 mg	12.8 mg
Selenium	8.4 µg	2.8 µg

### **3.4 Discussion**

As high concentrations of toxic metabolites accumulate in the culture medium, frequent changing of medium is required to maintain proliferation of hiPSC in conventional suspension culture. In this study, to reduce the frequency of changing medium, a refinement system was constructed with a dialysis filter to remove toxic metabolites and to recycle macromolecules (including autocrine factors) in the culture medium, after determining the minimum inhibitory levels of lactic acid for hiPSC. The refinement system removed lactic acid and ammonium from the used medium, supplied high concentrations of glucose, and retained bFGF and TGF- $\beta$ 1 in the refined medium after dialysis. The refined medium promoted the growth of hiPSC after supplying bFGF in the adherent culture, since most of the bFGF had been consumed by the cells. When the refinement system was applied to suspension culture, a high cell density as well as high levels of pluripotency markers expression was obtained.

In the present study, hiPSCs were observed to be more sensitive to lactic acid than other cell types (**Fig. 3.2**A) (Hassell *et al.*, 1991; Ozturk *et al.*, 1992; Lao and Toth, 1997; Patel *et al.*, 2000; Ouyang *et al.*, 2007; Schop *et al.*, 2009; Chen *et al.*, 2010; Luo *et al.*, 2012). As a consequence, 1 g L<sup>-1</sup> of lactic acid inhibited the growth of hiPSCs, and caused lower expression levels of pluripotency markers, although other types of cells can tolerate more than 1 g L<sup>-1</sup> of lactic acid. Human mesenchymal stem cells (hMSCs) and mouse hybridoma cells have been reported to tolerate more than 3 g L<sup>-1</sup> of lactic acid (Ozturk *et al.*, 1992; Patel *et al.*, 2009), whereas, human hematopoietic cells, Chinese hamster ovary (CHO) cells, and monkey kidney cells can tolerate about 2 g L<sup>-1</sup> of lactic acid (Hassell *et al.*, 1991; Lao and Toth, 1997; Patel *et al.*, 2000; Luo *et al.*, 2012). Moreover, lactic acid had direct effects on hiPSC as the growth rate

significantly decreased with the increase in lactic acid concentration, even though the initial pH of the medium had been adjusted. Ouyang *et al.* reported that 1.5 g  $L^{-1}$  or more of lactic acid had toxic effects on mouse embryonic stem cells (mESCs) and could induce spontaneous differentiation (Ouyang et al., 2007). In addition, Chen et al. reported that the frequency of a hESC pluripotency marker (TRA-1-60) was reduced significantly in cultures with high concentration of lactic acid (1 g L<sup>-1</sup>) (Chen et al., 2010). On the other hand, the specific production rate of lactic acid in hPSC was higher than it was in other types of cells. The specific production rate of lactic acid in mESC, hESC and hiPSC has been reported to be 0.09, 0.07, and 0.08 (present study) g  $h^{-1}$  10<sup>-9</sup> cells, respectively (Terstegge *et al.*, 2007; Chen *et al.*, 2010), whereas, it was 0.002, 0.04, 0.01, and 0.04 g h<sup>-1</sup> 10<sup>-9</sup> cells for CHO, hybridoma, hematopoietic, and Vero cells, respectively (Ozturk et al., 1992; Patel et al., 2000; Quesney et al., 2003; Luo et al., 2012). From this result, higher production of lactic acid was confirmed to be the major limiting factor for the proliferation of hiPSC, since hiPSC showed lactic acid toxicity at a lower concentration (1 g L<sup>-1</sup>) than in other types of cells. To reduce the toxic effects of lactic acid, frequent removal of lactic acid from culture medium is advantageous for the proliferation of hiPSC.

Our proposed medium refinement system was efficient in removing lactic acid from the used medium and maintaining a high concentration of glucose in the refined medium (**Table 3.1**). It also removed ammonium from the used medium, although only trace amounts of ammonium were produced by cells which did not exceed the minimum inhibitory level (Chen *et al.*, 2010). This was important because high concentrations of ammonium can be toxic to cells, causing cytosol vacuolization and subsequent cell death (Silvac *et al.*, 2010). Based on these results, it
appears that the medium refining system could replace conventional medium change for the expansion of hiPSC in suspension culture.

The retention of bFGF and TGF- $\beta$ 1 in the refined medium showed that the refining system allowed efficient recycling of costly macromolecules including autocrine factors (**Table 3.1**). The concentration of bFGF detected in the refined medium was lower than that in fresh medium (100 µg L<sup>-1</sup>) (Levenstein *et al.*, 2006; Ludwig *et al.*, 2006), and most of the bFGF was found to be consumed by cells (**Fig. 3.3**). Previously, it was assumed that thermal instability was the major cause for low concentrations of bFGF in medium after 24 h of culture, as bFGF can aggregate in an inactive form at 37°C (Chen *et al.*, 2012). By stabilizing bFGF with heparin (Furue *et al.*, 2008; Chen *et al.*, 2012), we confirmed that not only thermal aggregation; but also consumption by cells were the major causes of low bFGF concentrations in the used, 24 h suspension culture medium. However, a similar growth rate and expression of pluripotency markers in both fresh and refined medium supplemented with bFGF showed that the refined medium could be used effectively for the expansion of hiPS cells in adherent culture (**Fig. 3.4**).

As high concentrations of toxic metabolites accumulate during daily medium change in conventional suspension culture, Fernandes-Platzgummer *et al.* reported that periodic removal (every 12 h) of these metabolites was beneficial for cell growth when culturing mESCs in a microcarrier-immobilized stirred tank reactor (Fernandes-Platzgummer *et al.*, 2014). In this study, when medium was exchanged every 12 h after day 2, a final cell density of  $(8.1 \pm 2.2) \times 10^5$  cells mL<sup>-1</sup> with the expression of pluripotency markers were obtained on day 4 in conventional suspension culture (**Fig. 3.5**). Human iPSC cultured with the refining system, in

which DMEM/F12 was used as a dialysis medium, showed a similar cell density  $((1.1 \pm 0.1) \times 10^{6} \text{ cells mL}^{-1})$ , and metabolic activity when compared with the conventional suspension culture. In suspension culture with the medium refining system, low molecular weight components of the fresh medium (insulin (5.8 kDa), selenium (0.08 kDa), L-ascorbic acid (0.18 kDa), pipecolic acid (0.13 kDa), gamma-aminobutyric acid (GABA) (0.1 kDa), etc. in mTeSR<sup>TM</sup>1), which might pass through the dialysis membrane (MWCO: 10 kDa), were removed from the used culture medium. Only high molecular weight components (bFGF (18 kDa), TGF- $\beta$ 1 (25 kDa), transferrin (80 kDa), bovine serum albumin (BSA) (66 kDa) and autocrine factors) were retained in the refined medium after dialysis (**Table 3.3**).

Although insulin, selenium, and L-ascorbic acid play prominent roles in cell survival, selfrenewal, and proliferation of hPSCs during multiple passages (Chen *et al.*, 2011), the refined medium promoted the growth of hiPSCs without these components. It was assumed that the roles of these removed-components were fulfilled by the retained-components of the refined medium, especially bFGF and TGF- $\beta$ 1, as well as other autocrine factors that have overlapping functions of promoting self-renewal and undifferentiated proliferation of hPSCs (Amit *et al.*, 2004; James *et al.*, 2005; Xu *et al.*, 2005). During suspension culture with the medium refinement system, higher concentrations of bFGF and TGF- $\beta$ 1 were detected in the 96 h used culture medium, than that in the conventional suspension culture (**Fig. 3.6**). This remaining amount of bFGF and TGF- $\beta$ 1 were possibly due to less consumption by cells, or recycling by the medium refining system, or secretion from cells in suspension culture. hESCs were reported to secrete bFGF, and stimulate the production of TGF- $\beta$ 1 and insulin as autocrine factors in adherent culture (Dvorak *et al.*, 2005; Bendall *et al.*, 2007; Greber *et al.*, 2007; Montes *et al.*, 2009).

Medium components	Conventional culture	Medium refinement system
DMEM/F12	$\checkmark$	
L-Ascorbic acid	$\checkmark$	-
Selenium	$\checkmark$	-
Transferrin	$\checkmark$	*
NaHCO <sub>3</sub>	$\checkmark$	-
Glutathione	$\checkmark$	-
L-Glutamine	$\checkmark$	-
Defined lipids	$\checkmark$	-
Thiamine	$\checkmark$	-
Trace element B	$\checkmark$	-
Trace element C	$\checkmark$	-
BSA	$\checkmark$	*
bFGF	$\checkmark$	$\checkmark$
TGF-β1	$\checkmark$	$\checkmark$
Insulin	$\checkmark$	-
Pipecolic acid	$\checkmark$	-
LiCl	$\checkmark$	-
GABA	$\checkmark$	-
Autocrine factors		*

Table 3.3 Comparison of medium components between conventional culture and medium refinement system

√: Included in medium
\*: Expected to be contained in the refined medium
-: Expected not to be contained in the refined medium

Blank: Not included in medium

In the conventional suspension culture, autocrine factors are removed from culture medium during daily medium change. As suspension culture of hiPSC requires large volume of medium supplemented with high concentration of growth factors (macromolecules) (Tian *et al.*, 2004; Ludwig *et al.*, 2006), removal of autocrine factors during daily medium change results in high consumption of macromolecules. During culture with the medium refining system, additional supplementation with macromolecules was not needed, except for bFGF and TGF- $\beta$ 1 (50% less supplementation) (**Table 3.2**). Therefore, the established medium refinement system may provide a platform for saving growth factors for the expansion of hiPSCs by recycling macromolecules, as well as autocrine factors in suspension culture.

In conclusion, a culture medium refinement method by dialysis was established in which toxic metabolites were removed, and macromolecules of medium (bFGF, TGF- $\beta$ 1, etc.) including autocrine factors were recycled efficiently for the expansion of hiPSCs in suspension culture. As large numbers of cells with high levels of pluripotency markers expression were obtained by exchanging fresh medium only once, and further refinement of medium with the refinement system, this system is proposed to provide a cost-effective platform by reducing the consumption of growth factors for the expansion of hiPS cells in suspension culture.

# **3.5 Summary**

hiPSCs secrete essential autocrine factors that are removed along with toxic metabolites when the growth medium is exchanged daily. In this study, after determining the minimum inhibitory level of lactic acid for hiPSCs, a medium refinement system was constructed by which toxic metabolites were removed from used culture medium, and autocrine factors as well as other growth factors were recycled. Specifically, about 87% of bFGF and 80% of TGF- $\beta$ 1 were retained in the refined medium after dialysis. The refined medium efficiently potentiated the proliferation of hiPS cells in adherent culture. When the refinement system was used to refresh medium in suspension culture, a final cell density of  $(1.1 \pm 0.1) \times 10^6$  cells mL<sup>-1</sup> was obtained, with 99.5%  $\pm$  0.2% OCT 3/4 and 78.3%  $\pm$  1.1% TRA-1-60 expression, on day 4 of culture. These levels of expression were similar to those observed in conventional suspension culture. With this method, culture medium refinement by dialysis was established to remove toxic metabolites, recycle autocrine factors, as well as other growth factors, and reduce the use of macromolecules for the expansion of hiPSCs in suspension culture.

### **Chapter 4**

# Suspension culture of hiPSCs with medium refinement and aggregate breakup for high-density and high-yield

# **4.1 Introduction**

After the first isolation of human induced pluripotent stem cells (hiPSCs) by Yamanaka group (Takahashi *et al.*, 2007), application of these cells are increasing day by day as regenerative medicine for cell replacement therapies (Schulz *et al.*, 2012), tissue engineering (Badylak *et al.*, 2011), and drug discovery (Wernig *et al.*, 2007) due to their pluripotency and self-renewal capacity. To realize their potential, large-scale expansion of hiPSCs is important as practically more than  $10^9$ – $10^{10}$  cells per patient are needed for clinical application (Lock and Tzanakakis, 2007; Jing *et al.*, 2008; Kehoe *et al.*, 2010; Olmer, *et al.*, 2010; Serra *et al.*, 2010). Compared to adherent culture, suspension culture of hiPSC aggregate is promising to obtain such numbers of cells, as a higher cell density, and better control of process parameters can be achieved in a stirred suspension bioreactor culture (Zweigerdt *et al.*, 2011; Olmer *et al.*, 2012; Haraguchi *et al.*, 2015).

During the initial seeding in suspension culture, single hiPSCs are inoculated into culture vessel, and cells agglomerate to form sphere shape aggregates by encountering and connecting with neighboring cells which is mediated by E-cadherin (Sugawara *et al.*, 2010; Li *et al.*, 2012). The size of aggregate increases with the increase of culture time and some of the aggregates simultaneously fuse to form an un-uniform shape (Bauwens *et al.*, 2008; Otsuji *et al.*, 2014). The prolongation of culture time leads to undesired differentiation, as well as cell necrosis inside the aggregate which also cause the difficulty for maintaining proper growth in suspension culture (Sachlos *et al.*, 2008). Moreover, accumulation of toxic components increases with the increase

of cell density in suspension culture which inhibits the growth of hiPSC (Chen *et al.*, 2010). Production of lactic acid is higher in human pluripotent stem cell (hPSC) culture than it is in other cell types (Varum *et al.*, 2011), and lactic acid has potential toxic effects on the proliferation of hPSC (Ouyang *et al.*, 2007; Chen *et al.*, 2010). As a result, frequent changing of culture medium is required to remove toxic metabolites (Fernandes-Platzgummer *et al.*, 2014). On the other hand, hPSCs secrete various autocrine factors in the culture medium that maintain culture homeostasis, and potentiate the growth of cells (Bendall *et al.*, 2007; Montes *et al.*, 2009). These autocrine factors are removed from culture medium along with toxic metabolites during daily medium changes in conventional suspension culture. Moreover, the cost of culturing hiPSC increases with increasing medium consumption, as the culture medium contains many costly macromolecules including bFGF, TGF- $\beta$ 1, and insulin (Tian *et al.*, 2004; Ludwig *et al.*, 2006). Therefore, maintaining small size aggregates and recycling costlier macromolecules is important to obtain high cell density as well as for reducing the culture cost.

In this study, an improved operational design of hiPSC culture was applied for obtaining high cell density by breaking up of hiPSCs aggregates into small sizes, and integration of medium refinement system for efficient recycling of autocrine factors and other growth factors in suspension culture. This integrated system was also investigated for high-yield on medium in suspension culture.

## 4.2 Materials and methods

### 4.2.1 Culture of hiPSCs

hiPSCs were obtained from Japanese Collection of Research Bioresources (Tic, JCRB number: 1331). For subculture, single cells  $(1.0 \times 10^4 \text{ cells cm}^{-2})$  were seeded with 10 µM ROCK inhibitor (Tocris Bioscience, UK) in a 55 cm<sup>2</sup> culture dish (Corning Inc., USA) coated with laminin fragment (iMatrix<sup>TM</sup>-511, Nippi Inc., Japan), with hiPSC expansion medium, mTeSR<sup>TM</sup>1 (STEMCELL Technologies, Canada), 2 mm in depth. Cells were incubated at 37°C in a humidified atmosphere with 5% CO<sub>2</sub>, and medium was exchanged daily with fresh medium. On day 4, when cells reached 80–90% confluence, cells were subcultured. For routine passage, hiPSCs were treated with 5 mM EDTA/PBS for 7 min at room temperature with 10 µM ROCK inhibitor. After that, dissociation reagent (TrypLE Select<sup>TM</sup>, Invitrogen, USA) with10 µM ROCK inhibitor was applied for another 7 min at room temperature. After dissociating the hiPSC colony into single cells,  $1.0 \times 10^4$  cells cm<sup>-2</sup> were seeded with 10 µM ROCK inhibitor into a new culture dish.

### 4.2.2 Medium refinement system

The medium refinement system was designed with a cross flow dialysis membrane filter by which used suspension culture medium was refreshed by using DMEM/F12 (Sigma-Aldrich, USA) as the dialysate. The cross flow dialysis membrane filter contained 300 hollow fibers, and had a 200  $\mu$ m internal diameter, a MWCO of 10 kDa, and a surface area of 150 cm<sup>2</sup> (AKH-5S, Toyobo) (**Fig. 4.1**). Aseptic connections were maintained using silicone tubing. The flow rate of used medium was 2.6 mL min<sup>-1</sup>, whereas, the cross flow rate of fresh basal medium

(DMEM/F12), which was considered to be equivalent to the basal component in expansion medium, was 6.6 mL min<sup>-1</sup>.

### 4.2.3 Suspension culture with the medium refinement and aggregate break-up

Suspension culture of hiPSC aggregates was performed by seeding single cells at a density of  $2.0 \times 10^5$  cells mL<sup>-1</sup> in a stirred suspension bioreactor (working volume: 120 mL, BWV-S10A, Able Co., Japan) with ROCK inhibitor (Fig. 4.1). In the expansion medium, dissolved oxygen (DO, 3.2 mg L<sup>-1</sup>) was regulated by air and N<sub>2</sub> gas. Moreover, pH was regulated by CO<sub>2</sub> and the agitation rate was maintained at 40 rpm. In the control culture, medium was exchanged with fresh one every 24 h during the first 48 h and every 12 h thereafter. During culture with the medium refining system, spent medium was exchanged with fresh medium only at t = 24 h to remove the ROCK inhibitor. At t = 48 h, 100 mL of used medium was transferred to another bioreactor which was set with the medium refinement system. Further culture was performed by dialyzing the used medium with the basal medium (600 mL at each time) supplemented with bFGF (100  $\mu$ g L<sup>-1</sup>) every 12 h. After finishing the medium refinement at each time, the refined medium was transferred to the original bioreactor. At 87 h, HA (40 nM, because the aggregates were difficult to break with 20 nM in the refined medium) was added to the culture medium, and kept for 9 h. Aggregates were broken into small sizes by pipetting, and all of the cells were re-seeded into the bioreactor with 10 µM ROCK inhibitor and cultured for additional 96 h. In the control culture, aggregates were not treated with HA, as well as not broken into small sizes by pipetting.



**Fig. 4.1** Schematic diagram showing the experimental methods for integrating the medium refinement system with aggregate break-up by HA for obtaining high-density of hiPSCs in suspension culture.

### 4.2.4 Kinetic Analysis of Growth Properties of hiPSC Aggregates

To determine apparent specific growth rate,  $\mu^{app}$  (h<sup>-1</sup>), hiPSC aggregates were collected from the bioreactor (3 samples each time) at every 24 h, and dissociated into single cells by using TrypLE Select<sup>TM</sup>. After centrifugation, supernatant was discarded, and cells were re-suspended in fresh medium. Then live cells were counted by using the cell counter. Initial live cell ratio,  $\alpha$  (-), was determined by  $\alpha = X_{24}/X_0$ , where,  $X_0$  is the cell density at seeding, and  $X_{24}$  is the cell density at t = 24 h. Apparent specific growth rate,  $\mu^{app}$  (h<sup>-1</sup>), was calculated by the following equation:  $\mu^{app} = \ln(X_f/X_i)/(t_f - t_i)$ , where,  $X_f$  is the final cell density at  $t_f = 87$  or 192 h, and  $X_i$  is the initial cell density at  $t_i = 24$  or 120 h. Expansion fold (-) was calculated by  $X_f/X_i$ , where,  $X_f$  is the final cell density at  $t_f = 87$  or 192 h, and  $X_i$  is the initial cell density at  $t_i = 0$  or 96 h. Yield on culture medium, *Y* (cells mL<sup>-1</sup>) was calculated by the following equation,  $Y = N_t/V_m$ , where,  $N_t$  is the total number of cells (cells), and  $V_m$  is the total volume of medium consumption (mL). To determine the aggregate size, images were captured at every 24 h by using an image analyzer with a 4× objective lens (IN Cell Analyzer 2000; GE Healthcare, USA), and aggregate diameter was measured by using Image-Pro Plus software (Version 6.0; Media Cybernetics, USA).

### 4.2.5 Analysis of Medium Components

To analyze the components of culture medium associated with cell cultivation, medium was collected at every time during medium change, and analyzed for glucose and lactic acid by using a biochemical analyzer (Bioprofile 400; Nova Biomedical, USA). To determine the amounts of remaining growth factors, specifically bFGF and TGF- $\beta$ 1 in the culture medium, used medium was collected at *t* = 24, 87, and 192 h, and analyzed by an ELISA kit (R&D Systems, UK).

### 4.2.6 Flow Cytometry

To perform flow cytometry, cells were collected at t = 192 h for checking the expression of pluripotency markers, OCT 3/4 and SSEA 4. The procedure for flow cytometry was similar to that described previously. Briefly, after preparing single cells, cells were treated with the Cytofix/Cytoperm<sup>TM</sup> permeabilization kit (BD Biosciences, USA) and incubated with primary antibody, mouse monoclonal OCT 3/4 (Santa Cruz Biotechnology, USA), and secondary antibody, APC anti-mouse IgG (BD Biosciences). Direct staining of SSEA 4 was performed by using human/mouse SSEA 4 fluorescein (R&D Systems, USA). Flow cytometry was performed with a flow cytometer (Sysmex), and data were analyzed by commercially available software (FlowLogic, Inivai Technologies, Australia).

### 4.2.7 Fluorescence Staining

The procedure for staining was similar to that described previously. Briefly, hiPSC aggregates were washed with PBS, and then embedded in optical cutting temperature compound (Tissue-Tek; Sakura Fine Chemical Co., Ltd.), and sectioned into 20-µm- thick slices by using a cryostat microtome (Leica). The sections were then washed with PBS, and fixed with 4% paraformaldehyde (Wako Pure Chemical Industries) for 10 min at room temperature. After washing with PBS, the specimens were permeabilized with PBS containing 0.5% Triton X-100 (Wako Pure Chemical Industries) for 5 min, then washed twice with PBS containing 0.1% Tween 20 (Sigma-Aldrich), and blocked in Block Ace (DS Pharma Co. Ltd.) for 90 min at room temperature. The sections were washed with Tris-buffered saline (TBS; Dako), and then probed with anti-E-cadherin antibody (Santa Cruz Biotechnology, USA), and anti-collagen type I antibody (Abcam, USA), and kept overnight at 4°C. Next, they were washed twice with TBS, and immuno-labeled with Alexa Fluor 488-conjugated goat anti-rabbit (Santa Cruz Biotechnology, USA) or Alexa Fluor 594-conjugated goat anti-mouse IgG (Santa Cruz Biotechnology) for 60 min at room temperature. The cell nuclei were stained with 4', 6diamidino-2-phenylindole (DAPI; Life Technologies) for 20 min at room temperature. Images were captured by using a confocal laser scanning microscope (FV 1000; Olympus), with 20× and  $60 \times$  objective lens.

#### **4.2.8 Statistical Analysis**

Data presented in this study were obtained from two independent experiments and expressed as mean  $\pm$  standard deviation (SD). Statistical comparisons were evaluated using the Student's ttest and values of p < 0.01 and p < 0.05 were considered to be significant.

# **4.3 Results**

To obtain high-density of hiPSCs, aggregates were exposed to HA for 9 h, and broken up into small sizes at t = 96 h by pipetting. Lower expression of E-cadherin was observed after 9 h of HA addition which indicated the weakening of cell-cell connections by disrupting the E-cadherin located in the intercellular junctions during culture with the integrated system (**Fig. 4.2**). hiPSC aggregates showed higher expression of collagen type I at the periphery before HA addition at t = 87 h, whereas, the expression decreased at the aggregate periphery after exposure to HA at t = 96 h.



**Fig. 4.2** Fluorescent images of hiPSC aggregates for DAPI (blue), E-cadherin (red), and collagen type I (green) before (A-D), and after (E-H) adding HA at 87 and 96 h, respectively. Panels D and H show magnified views of the boxed areas. Scale bars: 100 µm.



Fig. 4.3 High-density culture of hiPSCs by aggregate break-up and medium refinement. (A) Growth profile, and (B) aggregate size of hiPSCs in the conventional culture without aggregate break-up (close circle), and integrated culture (open circle). Glucose consumption (open circle), and lactic acid production (close circle) in the (C) conventional culture without aggregate break-up, and (D) integrated culture with aggregate break-up. (E) Yield on culture medium. (F) Flow cytometry for OCT 3/4 and SSEA 4 in the conventional culture (open bar), and integrated culture (close bar). \*p < 0.01.

	Conventional culture		Integrated culture	
	Early-stage (0–96 h)	Late-stage (96–192 h)	Early-stage (0–96 h)	Late stage (96–192 h)
Apparent specific growth rate, $\mu^{app}$ (10 <sup>-2</sup> h <sup>-1</sup> )	$3.2 \pm 0.2$	1.0 ± 0.3	3.2 ± 0.1	$2.0 \pm 0.1*$
Final cell density, $X_{f}$ (10 <sup>5</sup> cells mL <sup>-1</sup> )	$9.2\pm0.5$	$16.3 \pm 1.9$	$1.3 \pm 0.1$	44.1 ± 3.5*
Expansion fold (-)	$4.5\pm0.2$	$1.5\pm0.1$	$6.5\pm0.5$	$3.3\pm0.1*$
Yield on culture medium (0-192 h) ( $10^4$ cells mL <sup>-1</sup> )	$0.1 \pm 0.01$		$2.1 \pm 0.02*$	
* <i>p</i> < 0.01				

**Table 4.1** Comparison of growth properties of hiPSCs in conventional culture and integrated culture.

hiPSC showed higher cell density (( $4.3 \pm 0.3$ ) × 10<sup>6</sup> cells mL<sup>-1</sup>) in the integrated culture in comparison to that in the conventional culture without aggregate break-up (**Fig. 4.3A**). When aggregates were broken into small sizes with HA, small size of aggregates was observed at *t* = 120 h, and the size of aggregates increased gradually thereafter (**Fig. 4.3B**). However, aggregate size did not change significantly during late-stage of conventional culture. Higher consumption of glucose and production of lactic acid was observed in hiPSC in the integrated culture compared to conventional culture (**Fig. 4.3C and D**). Lactic acid was not completely removed from the used medium in the integrated culture. However, hiPSC showed higher yield on culture medium in the integrated culture compared to the conventional culture during the late-stage (**Fig. 4.3 E**). Moreover, similar frequency of OCT 3/4 and SSEA 4 were observed in the integrated culture and the conventional culture without aggregate break-up (**Fig. 4.3 F**).



**Fig. 4.4** Fluorescent images of hiPSC aggregates for DAPI (blue) and collagen type I (green) after t = 192 h in conventional culture (A-C), and integrated culture (D-F). Panels C and F show magnified views of boxed areas. Scale bars: 100 µm.

hiPSCs showed higher specific growth rate  $((2.0 \pm 0.1) \times 10^{-2} h^{-1})$  in the integrated culture compared to the conventional culture without aggregate break-up  $(1.0 \pm 0.3) \times 10^{-2} h^{-1})$  during late-stage of culture (**Table 4.1**). In addition, more than 21-times higher expansion folds of hiPSCs were obtained by the integrated culture in comparison to the conventional culture without aggregate break-up. Higher accumulation of collagen type I was observed on the aggregate periphery at t = 192 h in the conventional culture in comparison to the integrated culture (**Fig. 4.4**).



Fig. 4.5 Remaining amount of cytokines in the used culture medium. Concentrations of (A) bFGF, and (B) TGF- $\beta$ 1 in the conventional culture (open bar), and integrated culture (close bar). \* p < 0.01.

Higher concentrations of bFGF was detected at t = 87, and t = 192 h in the used medium of integrated culture compared to the conventional culture (**Fig. 4.5A**). The concentration of TGF- $\beta$ 1 was similar in both conventional culture and culture with the integrated system during 24-87 h (**Fig. 4.5B**). However, the concentration of TGF- $\beta$ 1 decreased significantly at t = 192 h in the integrated culture. During expansion of hiPSCs with the integrated culture, additional supplementation of macromolecules (growth factors) was not needed, except for bFGF (**Table 4.2**). Therefore, supplementation of a large amount of costlier macromolecules was reduced, especially TGF- $\beta$ 1 and insulin during culture with the medium refinement system.

Required items	Conventional culture	Integrated culture
Medium		
Fresh medium (mTeSR <sup>TM</sup> 1) (L)	1.6	0.2
Dialysis medium (DMEM/F12) (L)		8.0
Medium components		
bFGF (µg)	160	160
TGF-β1 (μg)	3.2	0.6
Insulin (mg)	30.9	3.9
Transferrin (mg)	17.1	2.1
L-ascorbic acid (mg)	102.4	12.8
Selenium (µg)	22.4	2.8

**Table 4.2** Theoretical amounts of medium and other components required for conventional suspension culture and integrated culture for 100-mL-scale. The estimated amounts are for a single experiment.

# **4.4 Discussion**

In this study, an integrated culture was established by combining the medium refining system with the aggregate break-up by HA for obtaining high-density of hiPSCs with low supplementation of growth factors as shown in **Fig. 4.7**. The integrated culture was efficient in providing high yield on culture medium, as well as recycling the autocrine factors for reducing the consumption of growth factors in suspension culture.

In the integrated culture, after adding HA, compact aggregates with smooth edge turned into fragile aggregates with rough edge which was further broken into small sizes by pipetting after 9 h of HA addition (**Fig. 4.2**). Generally, cell-cell connections in hiPSC aggregates are mediated by E-cadherin (Chen *et al.*, 2010; Li *et al.*, 2012). While hiPSC aggregates were exposed to HA,

cell-cell connections were loosened by directly disrupting the E-cadherin located in the intercellular barrier of hiPSCs aggregate. Thus, lower expression of E-cadherin was observed throughout the aggregate after 9 h of HA addition (**Fig. 4.2G**). HA was reported to disrupt the E-cadherin mediated cell-cell connections in different types of cells in adherent culture (Sugawara *et al.*, 2010; Lee *et al.*, 2014; Kim *et al.*, 2017). Moreover, the distribution of collagen type I in the aggregate periphery changed significantly after 9 h of HA addition (**Fig. 4.2D and H**). While aggregates were exposed to HA, the E-cadherin mediated cell-cell connections were disrupted which was considered to promote higher migration of cells in the aggregate. High migration of cells might result in the disruption of organized collagen type I skeleton in the aggregate after 9 h of HA addition in the culture medium.



**Fig. 4.6** Comparison of final cell density and apparent specific growth rate of hiPSCs reported by other researchers (black circle) and current result obtained in this study (red circle) in stirred suspension bioreactor culture.



Fig. 4.7 Schematic interpretation of the current study.

Moreover, the fluid shear force in the bioreactor also contribute to wash out the collagen type I from the aggregate periphery which resulted in lower accumulation of collagen type I after 96 h in comparison to 87 h of HA addition in the culture medium.

Combining our previously established medium refinement system and aggregate break-up in the late-stage of culture resulted in high cell density which was three times higher than the control culture without aggregate break-up (**Fig. 4.3A**). Several recent studies have also been

reported to obtain high density of hPSCs  $(2.0 \times 10^6 \text{ cells mL}^{-1})$  in the conventional suspension culture (Amit *et al.*, 2010; Abbasalizadeh *et al.*, 2012; Olmer *et al.*, 2012) which are significantly lower density in comparison to the current study (**Fig. 4.6**). By using the integrated culture system, high cell density as well as high growth rate was obtained which is promising for largescale expansion of hiPSCs in suspension culture. Although hiPSCs can be expanded to a highdensity ( $3.6 \times 10^6$  cells mL<sup>-1</sup>) by continuous perfusion (Kropp *et al.*, 2016), maintaining such density in aggregate suspension culture is still challenging due to aggregate size limitation. Moreover, culturing hiPSCs for a prolonged period requires higher volume of medium which is significantly cost-intensive. By using our established integrated culture, high yield on medium was obtained by recycling costlier macromolecules as well as other autocrine factors to promote a cost-effective platform in suspension culture (**Fig. 4.3E**).

The proliferation of hiPSC aggregate reduced significantly during the late-stage of conventional culture due to large size of aggregates, and showed lower apparent specific growth rate in comparison to the integrated culture with aggregate break-up (**Table 4.1**). Large size of aggregates have been reported to show undesired differentiation, as well as cell necrosis in the center region of aggregate as collagen-rich shell forms on the outer surface of aggregate which hinders the diffusive transport of oxygen and nutrients during late-stage of culture (Sachlos *et al.*, 2008). In this study, lower growth rates in the conventional culture were considered due to higher accumulation of collagen type I on the aggregate periphery during late-stage of culture (**Fig. 4.4**). However, higher expansion fold of cells and higher yield on culture medium was obtained with the integrated system which was higher than those reported for hPSC expansion (2–4 folds) during long-term suspension culture (Abbasalizadeh *et al.*, 2012; Chen *et al.*, 2012; Olmer *et al.*, 2012; Hunt *et al.*, 2014).

Higher concentrations of bFGF was detected in the used medium of integrated culture at 87, and 192 h which was considered due to less consumption by cells, or recycling by the medium refinement system, or secretion from cells in suspension culture (**Fig. 4.5A**). hPSCs were reported to secrete bFGF, and stimulate the production of TGF- $\beta$ 1 and insulin as autocrine factors in adherent culture (Dvorak *et al.*, 2005; Bendall *et al.*, 2007; Greber *et al.*, 2007; Montes *et al.*, 2009). In this study, although TGF- $\beta$ 1 was not supplemented in the medium of integrated culture, higher concentration of TGF- $\beta$ 1 was detected at 87 h which was considered due to the secretion from cells as autocrine factors (**Fig. 4.5B**). Since additional supplementations with macromolecules (growth factors) was not needed in the integrated culture except for bFGF (**Table 4.2**), the established integrated culture provided a cost-effective platform for saving growth factors for the expansion of hiPSCs by recycling macromolecules, as well as other autocrine factors in suspension culture.

Moreover, the supplementation of TGF- $\beta$ 1 and heparin (for stabilization of bFGF) was not needed in this time which showed more cost-effective way of saving growth factors in comparison to the previous medium refinement system (described in chapter 3) (**Table 4.3**). This successful performance suggests the integrated culture as a promising platform for obtaining high cell density with high-yield on medium which will pave the way to establish an automatic and cost-effective platform for large-scale expansion of hiPSCs in suspension culture.

Medium components	Conventional culture	Medium refinement system (Chapter 3)	Integrated culture (Chapter 4)
DMEM/F12	$\checkmark$		$\checkmark$
L-Ascorbic acid	$\checkmark$	-	-
Selenium	$\checkmark$	-	-
Transferrin	$\checkmark$	*	*
NaHCO <sub>3</sub>	$\checkmark$	-	-
Glutathione	$\checkmark$	-	-
L-Glutamine	$\checkmark$	-	-
Defined lipids	$\checkmark$	-	-
Thiamine	$\checkmark$	-	-
Trace element B	$\checkmark$	-	-
Trace element C	$\checkmark$	-	-
BSA	$\checkmark$	*	*
bFGF	$\checkmark$	$\checkmark$	$\checkmark$
TGF-β1	$\checkmark$	$\checkmark$	-
Insulin	$\checkmark$	-	-
Pipecolic acid	$\checkmark$	-	-
LiCl	$\checkmark$	-	-
GABA		-	-
Autocrine factors		*	*
Heparin		$\checkmark$	

**Table 4.3** Comparison of medium components among the conventional culture, medium refinement system (Chapter 3), and integrated culture (Chapter 4).

 $\sqrt{1}$ : Included in medium, \* : Expected to be contained in the refined medium,

- : Expected not to be contained in the refined medium, Blank: Not included in medium

# 4.5 Summary

In this study, an integrated culture was established by combining the medium refinement system with aggregate break-up for obtaining high cell density with low supplementation of growth factors in suspension. The integrated culture showed high cell density which was three-times higher than the conventional culture without aggregate break-up. Moreover, higher expansion fold, as well as higher yield on culture medium (21-times) was obtained with the integrated culture in comparison to the conventional culture. Break-up of hiPSCs aggregates with HA during the late-stage of culture reduced the accumulation of collagen type I on the aggregate periphery in comparison to the conventional culture. Moreover, efficient recycling of bFGF and TGF-β1, as well as other autocrine factors provided a cost-effective platform by less supplementation of costlier macromolecules for obtaining high-density of hiPSCs in the integrated culture. This successful performance suggests the integrated culture as a promising platform for reducing the consumption of costlier macromolecules for large-scale expansion of hiPSCs in suspension culture.

# **General conclusion**

In this study, a simple and scalable method for high-density culture of hiPSCs was established by investigating the size- and time-dependent growth properties of aggregates, and controlling aggregate size during late-stage of culture by breaking up the aggregates into small sizes with HA, and reducing the supplementation of costlier macromolecules in suspension culture.

In Chapter 1, hiPSCs showed heterogeneous growth properties depending on aggregate size and culture time in single aggregate culture. Small size aggregates showed low growth rate due to low secretion of autocrine factors, whereas, large size aggregates showed low growth rate in comparison to medium size aggregates due to cellular quiescence during early-stage of culture. Culture time was also important for influencing heterogeneous growth during prolonged period by uneven distribution of inductive biochemicals because of the formation of an ECM-shell on the aggregate periphery. Therefore, considerations for aggregate size and culture time-dependent boundary conditions for maintaining the balance between autocrine factors secretion, and ECM accumulation on the aggregate periphery is important to obtain high growth rate of hiPSCs in suspension culture.

In Chapter 2, a simple method for *in situ* break-up of hiPSC aggregates by HA was established by elucidating the mechanism of E-cadherin mediated cell-cell connections disruption for controlling the aggregate size in suspension culture. Since large size aggregates show low growth rate during late-stage of culture, aggregates broken into small sizes with HA resulted in high cell density in suspension culture. The temporal activity of HA for disrupting E-cadherin mediated cell-cell connections facilitated the aggregates break-up into small sizes for minimizing the initial cell loss during seeding in suspension culture. This promising method of hiPSC aggregate break-up with HA paved the way to obtain high cell density with high growth rate by simplifying the culture operations without the need for enzymatic treatment or centrifugation in suspension culture.

In Chapter 3, more than 1.0 g L<sup>-1</sup> of lactic acid was found to be toxic to hiPSCs growth and pluripotency. The medium refinement system was efficient in removing the toxic metabolites including lactic acid and ammonium in suspension culture. This method also allowed recycling of macromolecules (bFGF, TGF- $\beta$ 1, insulin, etc.) including other autocrine factors for the expansion of hiPSCs in suspension culture. As large numbers of cells with high levels of pluripotency markers expression were obtained by exchanging fresh medium only once, and further refinement of medium with the refining system, this system provided a cost-effective platform by reducing the supplementation of growth factors for the expansion of hiPS cells in suspension culture.

In Chapter 4, the medium refinement system was integrated with aggregate break-up during the late-stage of culture which resulted in more than three-times higher cell density than the conventional culture without aggregate break-up in suspension culture. Higher expansion fold of hiPSCs, as well as higher yield on culture medium was obtained with the integrated culture in comparison to the conventional culture. hiPSCs aggregates broken with HA during the late-stage of culture reduced the accumulation of collagen type I on the aggregate periphery in comparison to the conventional culture without aggregate break-up. Furthermore, efficient recycling of bFGF and TGF- $\beta$ 1, as well as other autocrine factors helps to reduce the supplementation of costlier macromolecules in the integrated culture which provided a cost-effective platform for obtaining high-density of hiPSCs in suspension culture.

Overall, a simple and scalable method for high-density culture of hiPSCs was established by revealing the heterogeneous growth of hiPSCs, and setting up a boundary condition for aggregate size and culture time in suspension. The temporal activity of HA for disrupting E-cadherin mediated cell-cell connections paved the way for controlling the aggregate size, and establishing an in situ simple method of hiPSC aggregate break-up without initial cell loss during seeding in suspension culture. High cell density and high growth rate was also obtained by breaking up the aggregates into small sizes during the late-stage of culture in comparison to that in the conventional culture without aggregate break-up. This promising method of aggregate break-up provided the simplification of culture operations without enzymatic treatment or centrifugation in suspension culture. Moreover, the medium refinement system was efficient in removing toxic metabolites, as well as recycling bFGF, TGF- $\beta$ 1, and other autocrine factors which helps to reduce the supplementation of costlier macromolecules in suspension culture. Finally, the integration of medium refinement system with aggregate break-up facilitated high-density of hiPSCs with high-yield on culture medium by reducing the supplementation of growth factors in suspension culture. Thus, the established integrated culture provided a cost-effective platform for obtaining high cell density by efficient recycling of autocrine factors, and simplifying the culture operations for large-scale expansion of hiPSCs in suspension culture.

# **Proposals for future works**

To extend the current findings towards future application, the following proposals should also be considered:

### 1. Development of an efficient medium collecting filter

The current medium collecting filter used in the medium refinement system (in chapter 3) was not efficient for high-density culture. As a result, the medium collecting filter was excluded during high-density culture with the integrated system (in chapter 4). Although developing circular filter which can be submerged in the culture medium containing aggregate is difficult, further attempts should be taken to choose suitable materials for making filter which can prevent cell attachment on the filter also. Developing a more efficient medium collecting filter will promote better medium refinement in suspension culture.

### 2. Automation of integrated system for hiPSCs expansion and differentiation

The integration of medium refinement system with aggregate break-up by HA seems to be promising for the establishment of an automatic integrated system for hiPSCs expansion and differentiation in suspension culture. Since the medium refinement system can be automated by using glucose and lactic acid sensors, periodic removal of toxic metabolites is possible to perform automatically in suspension culture. Moreover, spontaneous break-up of aggregates is also possible after treating with HA, and optimizing the agitation rate of impeller in bioreactor, which will help to maintain small size of aggregates during late-stage of culture. Since the medium refinement system can keep the autocrine factors in culture medium, hiPSCs cultured by the integrated system may show high differentiation potential in comparison to the conventional culture. After obtaining a high cell density, differentiation of these cells will result in high number of desired cells in suspension culture which will lead to a cost-effective platform for large-scale differentiation of hiPSCs by the automatic integrated system.

# Nomenclature

t	Culture time	[h]
X	Cell density	[cells mL <sup>-1</sup> ]
$X_{ m i}$	Initial cell density	[cells mL <sup>-1</sup> ]
$X_0$	Seeding density	[cells mL <sup>-1</sup> ]
$X_{ m f}$	Final cell density	[cells mL <sup>-1</sup> ]
$X_{\mathrm{t}}$	Cell density at <i>t</i> h	[cells mL <sup>-1</sup> ]
$X_{24}$	Cell density at 24 h	[cells mL <sup>-1</sup> ]
$X_{48}$	Cell density at 48 h	[cells mL <sup>-1</sup> ]
<i>X</i> <sub>72</sub>	Cell density at 72 h	[cells mL <sup>-1</sup> ]
$X_{96}$	Cell density at 96 h	[cells mL <sup>-1</sup> ]
$X_{120}$	Cell density at 120 h	[cells mL <sup>-1</sup> ]
<i>X</i> <sub>192</sub>	Cell density at 192 h	[cells mL <sup>-1</sup> ]
$N_0$	Initial cell number	[cells]
$N_{24}$	Cell number at 24 h	[cells]
N <sub>72</sub>	Cell number at 72 h	[cells]
α	Viable cell ratio	[-]
$\mu^{ ext{app}}$	Apparent specific growth rate	$[h^{-1}]$
Nt	Total number of cells	[cells]
Y	Yield	[cells mL <sup>-1</sup> ]
Vm	Total volume of medium	mL
D	Diameter	μm

# Abbreviations

2D	Two-dimensional
3D	Three-dimensional
bFGF	Basic fibroblast growth factor
TGF-β1	Transforming growth factor beta 1
IGF	Insulin-like growth factors
BSA	Bovine serum albumin
DMEM	Dulbecco's modified Eagle's medium
ECM	Extracellular matrix
FBS	Fetal bovine serum
TBS	Tris-buffered saline
PBS	Phosphate buffered saline
SD	Standard deviation
CLSM	Confocal laser scanning microscopy
HLA	Human leukocytes antigen
APC	Allophycocyanin
СМ	Conditioned medium
SSEA	Stage-specific embryonic antigen
OCT	Octamer-binding transcription factor
DAPI	4',6-diamidino-2-phenylindole
EDTA	Ethylenediaminetetraacetic acid
ELISA	Enzyme-linked immunosorbent assay

# Literature cited

- Abbasalizadeh, S., Larijani, M. R., Samadian, A., Baharvand, H.: Bioprocess development for mass production of size-controlled human pluripotent stem cell aggregates in stirred suspension bioreactor. *Tissue Eng. Part C Meth.*, **18**, 831–851 (2012).
- Amit, M., Chebath, J., Margulets, V., Laevsky, I., Miropolsky, Y., Shariki, K., Peri, M., Blais, I., Slutsky, G., Revel, M., Itskovitz-Eldor, J.: Suspension culture of undifferentiated human embryonic and induced pluripotent stem cells. *Stem Cell Rev. Rep.*, 6, 248–259 (2010).
- Amit, M., Shariki, C., Margulets, V., Itskovitz-Eldor, J.: Feeder layer- and serum-free culture of human embryonic stem cells. *Biol. Reprod.*, **70**, 837-845 (2004).
- Badylak, S. F., Taylor, D., Uygun, K.: Whole-organ tissue engineering: decellularization and recellularization of three-dimensional matrix scaffolds. *Annu. Rev. Biomed. Eng.*, **13**, 27–53 (2011).
- Balzac, F., Avolio, M., Degani, S., Kaverina, I., Torti, M., Silengo, L., Small, J. V., Retta, S. F.:
  E-cadherin endocytosis regulates the activity of Rap1: a traffic light GTPase at the crossroads between cadherin and integrin function. *J. Cell Sci.*, **118**, 4765–4783 (2005).
- Bauwens, C. L., Peerani, R., Niebruegge, S., Woodhouse, K. A., Kumacheva, E., Husain, M., Zandstra, P. W.: Control of human embryonic stem cell colony and aggregate size heterogeneity influences differentiation trajectories. *Stem Cells*, **26**, 2300-2310 (2008).
- Beers, J., Gulbranson, D. R., George, N., Siniscalchi, L. I., Jones J., Thomson J. A., Chen G.: Passaging and colony expansion of human pluripotent stem cells by enzyme-free dissociation in chemically defined culture conditions. *Nat. Proto.*, 7, 2029-2040 (2012).
- Bendall, S. C., Stewart, M. H., Menendez, P., George, D., Vijayaragavan, K., Werbowetski-Ogilvie, T., Ramos-Mejia, V., Rouleau, A., Yang, J., Bossé, M., Lajoie, G., Bhatia, M.: IGF

and FGF cooperatively establish the regulatory stem cell niche of pluripotent human cells in vitro. *Nature*, **448**, 1015-1021 (2007).

- Burridge, P. W., Anderson, D., Priddle, H., Barbadillo, M. M. D, Chamberlain, S., Allegrucci, C., Young, L. E., and Denning, C.: Improved human embryonic stem cell embryoid body homogeneity and cardiomyocyte differentiation from a novel V-96 plate aggregation system highlights interline variability. *Stem Cells*, 25, 929-938 (2007).
- Carmeliet, P., Dor, Y., Herbert, J. M., Fukumura, D., Brusselmans, K., Dewerchin, M., Neeman, M., Bono, F., Abramovitch, R., Maxwell, P., Koch, C. J., Ratcliffe, P., Moons, L., Jain, R. K., Collen, D., Keshert, E.: Role of HIF-1alpha in hypoxia-mediated apoptosis, cell proliferation and tumor angiogenesis. *Nature*, **394**, 485-490 (1998).
- Chambers, S. M., Fasano, C. A., Papapetrou, E. P., Tomishima, M., Sadelain M., Studer, L.: Highly efficient neural conversion of human ES and iPS cells by dual inhibition of SMAD signaling. *Nat. Biotechnol.*, 27, 275–280 (2009).
- Chang, B., Choi, Y., Kim, J.: Collagen complexes increase the efficiency of iPS cells generated using fibroblasts from adult mice. *J. Reprod. Dev.*, **61**, 145–153 (2015).
- Chen, G., Gulbranson, D. R., Hou, Z., Bolin, J. M., Ruotti, V., Probasco, M. D., Smuga-Otto, K., Howden, S. E., Dio, N. R., Propson, N. E., Wagner, R., Lee, G. O., Antosiewicz-Bourget, J., Teng, J. M., Thomson, J. A.: Chemically defined conditions for human iPSC derivation and culture. *Nat. Meth.*, 8, 424-429 (2011).
- Chen, G., Gulbranson, D. R., Yu, P., Hou, Z., Thomson, J. A.: Thermal stability of fibroblast growth factor protein is a determinant factor in regulating self-renewal, differentiation, and reprogramming in human pluripotent stem cells. *Stem Cells*, **30**, 623-630 (2012).

- Chen, T., Yuan, D., Wei, B., Jiang, J., Kang, J., Ling, K., Gu, Y., Li, J., Xiao, L., Pei, G.: Ecadherin-mediated cell–cell contact is critical for induced pluripotent stem cell generation. *Stem Cells*, 28, 1315–1325 (2010).
- Chen, X., Chen, A., Woo, T. L., Choo, A. B., Reuveny, S., Oh, S. K.: Investigations into the metabolism of two-dimensional colony and suspended microcarrier cultures of human embryonic stem cells in serum-free media. *Stem Cells Dev.*, **19**, 1781-1792 (2010).
- Dvorak, P., Dvorakova, D., Koskova, S., Vodinska, M., Najvirtova, M., Krekac, D., Hamp, A.: Expression and potential role of fibroblast growth factor 2 and its receptors in human embryonic stem cells. *Stem Cells*, **23**, 1200-1211 (2005).
- Elanzew, A., Sommer, A., Pusch-Klein, A., Brüstle O., Haupt S.: A reproducible and versatile system for the dynamic expansion of human pluripotent stem cells in suspension. *Biotechnol.* J., 10, 1589–1599 (2015).
- Fernandes-Platzgummer, A., Diogo, M. M., da Silva, C. L., Cabral, J. M. S.: Maximizing mouse embryonic stem cell production in a stirred tank reactor by controlling dissolved oxygen concentration and continuous perfusion operation. *Biochem. Eng. J.*, 82, 81–90 (2014).
- Furue, M. K., Na, J., Jackson, J. P., Okamoto, T., Jones, M., Baker, D., Hata, R. I., Moore, H. D., Sato, J. D., Andrews, P. W.: Heparin promotes the growth of human embryonic stem cells in a defined serum-free medium. *Proc. Natl. Acad. Sci. USA*, **105**, 13409-13414 (2008).
- Greber, B., Lehrach, H., Adjaye, J.: Fibroblast growth factor 2 modulates transforming growth factor beta signaling in mouse embryonic fibroblasts and human ESCs (hESCs) to support hESC self-renewal. *Stem Cells*, **25**, 455-464 (2007).

- Haraguchi, Y., Matsuura, K., Shimizu, T., Yamato, M., Okano, T.: Simple suspension culture system of human iPS cells maintaining their pluripotency for cardiac cell sheet engineering. *J. Tissue Eng. Regen. Med.*, 9, 1363–1375 (2013).
- Hassell, T., Gleave, S., Butler, M.: Growth inhibition in animal cell culture: the effect of lactate and ammonia. *Appl. Biochem. Biotechnol.*, **30**, 29-41 (1991).
- Hunt, M. M., Meng, G., Rancourt, D. E., Gates, I. D., Kallos, M. S.: Factorial experimental design for the culture of human embryonic stem cells as aggregates in stirred suspension bioreactors reveals the potential for interaction effects between bioprocess parameters. *Tissue Eng. Part C Meth.*, 20, 76–89 (2014).
- Hwang, Y. S., Chung, B. G., Ortmann, D., Hattori, N., Moeller, H. C., Khademhosseini, A.: Microwell-mediated control of embryoid body size regulates embryonic stem cell fate via differential expression of WNT5a and WNT11. *Proc. Natl. Acad. Sci. USA*, **106**, 16978– 16983 (2009).
- Ishikawa, Y., Ito, S., Nagata, K., Sakai, L. Y., Bächinger H. P.: Intracellular mechanisms of molecular recognition and sorting for transport of large extracellular matrix molecules. *Proc. Natl. Acad. Sci. USA*, **113**, E6036-E6044 (2016).
- Jain R., K., Au P., Tam, J., Duda, D. G., Fukumura, D.: Engineering vascularized tissue. Nat. Biotechnol., 23, 821-823 (2005).
- James, D, Levine, A. J., Besser, D., Hemmati-Brivanlou, A.: TGFbeta/activin/nodal signaling is necessary for the maintenance of pluripotency in human embryonic stem cells. *Development*, 132, 1273-1282 (2005).
- Jenkins M. J. and Farid, S. S.: Human pluripotent stem cell-derived products: Advances towards robust, scalable and cost-effective manufacturing strategies. *Biotechnol. J.*, **10**, 83–95 (2015).

- Jing, D., Parikh, A., Canty, J. M., Tzanakakis, E. S.: Stem cells for heart cell therapies. *Tissue Eng. Part B Rev.*, **14**, 393–406 (2008).
- Joza, N., Susin, S. A., Daugas, E., Stanford, W. L., Cho, S. K., Li, C. Y., Sasaki, T., Elia, A. J., Cheng, H. Y., Ravagnan, L., Ferri, K. F., Zamzami, N., Wakeham, A., Hakem, R., Yoshida, H., Kong, Y. Y., Mak, T. W., Zúñiga-Pflücker, J. C., Kroemer, G., Penninger, J. M.: Essential role of the mitochondrial apoptosis-inducing factor in programmed cell death. *Nature*, 410, 549-554 (2001).
- Kehoe, D. E., Jing, D., Lock, L. T., Tzanakakis, E. S.: Scalable Stirred-suspension Bioreactor Culture of Human Pluripotent Stem Cells. *Tissue Eng. Part A*, **16**, 405–421 (2010).
- Kim, M. H., Kino-oka, M.: Maintenance of undifferentiated state of human induced pluripotent stem cells through cytoskeleton-driven force acting to secreted fibronectin on a dendrimerimmobilized surface. J. biosci. Bioeng., 118, 716-722 (2014).
- Kim, M. H., Sugawara, Y., Fujinaga, Y., Kino-oka, M.: Botulinum hemagglutinin-mediated selective removal of cells deviating from the undifferentiated state in hiPSC colonies. *Sci. Rep.*, 7, 93 (2017).
- Kimbrel, E. A., and Lanza, R.: Current status of pluripotent stem cells: moving the first therapies to the clinic. *Nat. Rev. Drug Disc.*, **14**, 681–692 (2015).
- Kinney, M., Sargent, C., McDevitt, T.: The multiparametric effects of hydrodynamic environments on stem cell culture. *Tissue Eng. Part B*, **17**, 249–262 (2011).
- Kropp, C., Kempf, H., Halloin, C., Robles-Diaz, D., Franke, A., Scheper, T., Kinast, K., Knorpp, T., Joos, T. O., Haverich, A., Martin, U., Zweigerdt, R., Olmer, R.: Impact of feeding strategies on the scalable expansion of human pluripotent stem cells in single-use stirred tank bioreactors. *Stem Cells Transl. Med.*, **5**, 1289-1301 (2016).
- Kiyoshi, O., Fujiki, A., Yanagihara, K., Tachikawa S., Hayashi Y., Ito, Y., Onuma, Y., Chan, T., Michiue, T., Furue, M. K., Asashima, M.: Enzyme-free passage of human pluripotent stem cells by controlling divalent cations. *Sci. Rep.*, **4**, 4646 (2014).
- Lao, M. S., and Toth, D.: Effects of ammonium and lactate on growth and metabolism of a recombinant Chinese hamster ovary cell culture. *Biotechnol. Prog.*, **13**, 688-691(1997).
- Lee, K., Zhong, X., Gu S., Kruel M. A., Dorner M. B., Perry K., Rummel A., Dong M., and Jin R.: Molecular basis for disruption of E-cadherin adhesion by botulinum neurotoxin A complex. *Science*, **344**, 1405–1410 (2014).
- Levenstein, M. E., Ludwig, T. E., Xu, R. H., Llanas, R. A., VanDenHeuvel-Kramer, K., Manning, D., Thomson, J. A.: Basic fibroblast growth factor support of human embryonic stem cell self-renewal. *Stem Cells*, 24, 568-574 (2006).
- Lei Y., Jeong D., Xiao J., Schaffer D. V.: Developing defined and scalable 3D culture systems for culturing human pluripotent stem cells at high densities. *Cell. Mol. Bioeng.*, 7, 172–183 (2014).
- Li, L., Bennett, S., Wang, L.: Role of E-cadherin and other cell adhesion molecules in survival and differentiation of human pluripotent stem cells. *Cell Adh. Mig.*, **6**, 59–70 (2012).
- Lock, L. T., Tzanakakis, E. S.: Stem/progenitor cell sources of insulin-producing cells for the treatment of diabetes. *Tissue Eng.*, **13**, 1399–1412 (2007).
- Ludwig, T. E., Bergendahl, V., Levenstein, M. E., Yu, J., Probasco, M. D., Thomson, J. A.: Feeder-independent culture of human embryonic stem cells. *Nat. Meth.*, **3**, 637-646 (2006).
- Luo, J., Vijayasankaran, N., Autsen, J., Santuray, R., Hudson, T., Amanullah, A., Li, F.: Comparative metabolite analysis to understand lactate metabolism shift in Chinese hamster ovary cell culture process. *Biotechnol. Bioeng.*, **109**, 146-156 (2012).

- Matsuura, K., Wada, M., Shimizu, T., Haraguchi, Y., Sato, F., Sugiyama, K., Konishi, K., Shiba,Y., Ichikawa, H., Tachibana, A., Ikeda, U., Yamato, M., Hagiwara, N., Okano, T.: Creation of human cardiac cell sheets using pluripotent stem cells. *Biochem. Biophys. Res. Commun.*, 425, 321-327 (2012).
- Miyazaki, T., Futaki, S., Suemori, H., Taniguchi, Y., Yamada, M., Kawasaki, M., Hayashi, M., Kumagai, H., Nakatsuji, N., Sekiguchi, K., and Kawase, E.: Laminin E8 fragments support efficient adhesion and expansion of dissociated human pluripotent stem cells. *Nat. Commun.*, 3, 1236 (2012).
- Miranda C. C., Fernandes T. G., Diogo M. M., Cabral J. M. S.: Scaling up a chemically-defined aggregate-based suspension culture system for neural commitment of human pluripotent stem cells. *Biotechnol. J.*, 11, 1628-1638 (2016).
- Mohr, J. C., Zhang, J., Azarin, S. M., Soerens, A. G., de Pablo, J. J., Thomson, J. A., Lyons, G. E., Palecek, S. P., and Kamp, T. J.: The microwell control of embryoid body size in order to regulate cardiac differentiation of human embryonic stem cells. *Biomaterials*, 31, 1885–1893 (2010).
- Montes, R., Ligero, G., Sanchez, L., Catalina, P., Cueva, T., Nieto, A., Melen, G. J., Rubio, R., García-Castro, J., Bueno, C., Menendez, P.: Feeder-free maintenance of hESCs in mesenchymal stem cell-conditioned media: distinct requirements for TGF-beta and IGF-II. *Cell Res.* 19, 698-709 (2009).
- Moon, S. H., Ju, J., Park, S. J., Bae, D., Chung, H. M., Lee, S. H.: Optimizing human embryonic stem cells differentiation efficiency by screening size-tunable homogenous embryoid bodies. *Biomaterials*, 35, 5987-5997 (2014).

- Moon, S. H., Lee, S. G., Cho, Y. J., Hong, K. S., Lee, J. H., Lee, H. J., Chung, H. M.: Assessment of differentiation aspects by the morphological classification of embryoid bodies derived from human embryonic stem cells. *Stem Cells Dev.*, **20**, 1925–1935 (2011).
- Ying N., Walsh P., Clarke D. L., Rowley J. A., Fellner T.: Scalable Passaging of Adherent Human Pluripotent Stem Cells. *Plos One*, **9**, e88012 (2014).
- Novosel, E. C., Kleinhans, C., Kluger P. J.: Vascularization is the key challenge in tissue engineering. *Adv. Drug Deliv. Rev.*, **63**, 300-311 (2011).
- Ohgushi, M., Matsumura, M., Eiraku, M., Murakami, K., Aramaki, T., Nishiyama, A., Muguruma, K., Nakano, T., Suga, H., Ueno, M., Ishizaki, T., Suemori, H., Narumiya, S., Niwa, H., Sasai, Y.: Molecular pathway and cell state responsible for dissociation-induced apoptosis in human pluripotent stem cells. *Cell Stem Cell*, **7**, 225-239 (2010).
- Olmer, R., Lange, A., Selzer, S., Kasper, C., Haverich, A, Martin U., Zweigerdt, R.: Suspension culture of human pluripotent stem cells in controlled, stirred bioreactors. *Tissue Eng. Part C Meth.*, **18**, 772–784 (2012).
- Olmer, R., Haase, A., Merkert, S., Cui, W., Palecek, J., Ran, C., Kirschning, A., Scheper, T., Glage, S., Miller, K., Curnow, E. C., Hayes, E. S., Martin, U.: Long term expansion of undifferentiated human iPS and ES cells in suspension culture using a defined medium. *Stem Cell Res.*, 5, 51-64 (2010).
- Ota, M., Sasaki, H.: Mammalian Tead proteins regulate cell proliferation and contact inhibition as a transcriptional mediator of Hippo signaling. *Development*, **135**, 4059-4069 (2008).
- Otsuji, T. G., Bin, J., Yoshimura, A., Tomura, M., Tateyama, D., Minami, I., Yoshikawa, Y., Aiba, K., Heuser, J. E., Nishino, T., Hasegawa, K., Nakatsuji, N.: A 3D sphere culture

system containing functional polymers for large-scale human pluripotent stem cell production. *Stem Cell Rep.*, **2**, 1-12 (2014).

- Ouyang, A., Robin, N., Yang, S. T.: Long-term culturing of undifferentiated ESCs in conditioned media and three-dimensional fibrous matrices without extracellular matrix coating. *Stem Cells*, **25**, 447-454 (2007).
- Ozturk, S. S., Riley, M. R., Palsson, B.: Effects of ammonia and lactate on hybridoma growth, metabolism, and antibody production. *Biotechnol. Bioeng.*, **39**, 418-431(1992).
- Patel, S. D., Papoutsakis, E. T., Winter, J. N., Miller, W. M.: The lactate issue revisited: Novel feeding protocols to examine inhibition of cell proliferation and glucose metabolism in hematopoietic cell cultures. *Biotechnol. Prog.*, 16, 885-892 (2000).
- Quesney, S., Marc, A., Gerdil, C., Gimenez, C., Marvel, J., Richard, Y., Meginier, B.: Kinetics and metabolic specificities of Vero cells in bioreactor cultures with serum free medium. *Cytotechnology*, **42**, 1–11 (2003).
- Sachlos, E., Auguste, D. T.: Embryoid body morphology influences diffusive transport of inductive biochemicals: A strategy for stem cell differentiation. *Biomaterials*, **29**, 4471-4480 (2008).
- Schulz, T. C., Young, H. Y., Agulnick, A. D., Babin, M. J., Baetge, E. E., Bang, A. G., Bhoumik, A., Cepa, I., Cesario, R. M., Haakmeester, C., Kadoya, K., Kelly, J. R., Kerr, J., Martinson, L. A., McLean, A. B., Moorman, M. A., Payne, J. K., Richardson, M., Ross, K. G., Sherrer, E. S., Song, X., Wilson, A. Z., Brandon, E. P., Green, C. E., Kroon, E. J., Kelly, O. G., D'Amour, K. A., Robins, A. J.: A scalable system for production of functional pancreatic progenitors from human embryonic stem cells. *PLoS One*, 7, e37004 (2012).

- Serra, M., Brito, C., Sousa, M. F., Jensen, J., Tostões, R., Clemente, J., Strehl, R., Hyllner, J., Carrondo, M. J., Alves, P. M.: Improving expansion of pluripotent human embryonic stem cells in perfused bioreactors through oxygen control. *Biotechnol. J.*, **148**, 208-215 (2010).
- Singh, H., Mok, P., Balakrishnan, T., Rahmat, S. N., Zweigerdt, R.: Upscaling single cellinoculated suspension culture of human embryonic stem cells. *Stem Cell Res.*, 4, 165–179 (2010).
- Slivac, I., Blajić, V., Radošević, K., Kniewald, Z., Gaurina S. V.: Influence of different ammonium, lactate and glutamine concentrations on CCO cell growth. *Cytotechnology*, **62**, 585-594 (2010).
- Son, M. Y., Kim H., Kim M., Cho Y. S.: Physical Passaging of Embryoid Bodies Generated from Human Pluripotent Stem Cells. *PLoS One*, **6**, e19134 (2011).
- Sugawara, Y., Matsumura, T., Takegahara, Y., Jin, Y., Tsukasaki, Y., Takeichi, M., Fujinaga,
  Y.: Botulinum hemagglutinin disrupts the intercellular epithelial barrier by directly binding
  E-cadherin. J. Cell Biol., 189, 691–700 (2010).
- Sugawara, Y., Yutani, M., Amatsu, S., Matsumura, T., Fujinaga, Y.: Functional dissection of the *Clostridium botulinum* Type B hemagglutinin complex: identification of the carbohydrate and E-cadherin binding sites. *PLoS One*, 9, e111170 (2014).
- Suh, H. N., Han, H. J.: Collagen I regulates the self-renewal of mouse embryonic stem cells through α2β1 integrin- and DDR1-dependent Bmi-1. J. Cell Physiol., 226, 3422–3432 (2011).
- Takahashi, K., Tanabe, K., Ohnuki, M., Narita, M., Ichisaka, T., Tomoda, K., Yamanaka, S.:
  Induction of pluripotent stem cells from adult human fibroblasts by defined factors. *Cell*, 131, 861-872 (2007).

- Terstegge, S., Laufenberg, I., Pochert, J., Schenk, S., Itskovitz-Eldor, J., Endl, E. B. O: Automated maintenance of embryonic stem cell cultures. *Biotechnol. Bioeng.*, 96, 195-201 (2007).
- Tian, X., Morris, J. K., Linehan, J. L., Kaufman, D. S.: Cytokine requirements differ for stroma and embryoid body mediated hematopoiesis from human embryonic stem cells. *Exp. Hematol.*, **32**, 1000-1009 (2004).
- Varum, S., Rodrigues, A. S., Moura, M. B., Momcilovic, O., Easley, C. A., Ramalho, S. J., Houten, B. V., Schatten, G.: Energy metabolism in human pluripotent stem cells and their differentiated counter parts. *PLoS One*, 6, e20914 (2011).
- Wada, K., Itoga, K., Okano, T., Yonemura, S., Sasaki, H.: Hippo pathway regulation by cell morphology and stress fibers. *Development*, **138**, 3907-3914 (2011).
- Wang, Y., Chou, B. K., Dowey, S., He, C., Gerecht, S., Cheng, L.: Scalable expansion of human induced pluripotent stem cells in the defined xeno-free E8 medium under adherent and suspension culture conditions. *Stem Cell Res.*, **11**, 1103–1116 (2013).
- Watanabe, K., Ueno M., Kamiya D., Nishiyama A., Matsumura M., Wataya T., Takahashi J. B., Nishikawa S., Muguruma K., Sasai Y.: A ROCK inhibitor permits survival of dissociated human embryonic stem cells. *Nat. Biotechnol.*, 25, 681–686 (2007).
- Wernig, M., Meissner, A., Foreman, R., Brambrink, T., Ku, M., Hochedlinger, K., Bernstein, B.
  E., Jaenisch, R.: In vitro reprogramming of fibroblasts into a pluripotent ES-cell-like state. *Nature*, 448, 318-324 (2007).
- Winkle, A. P. V., Gates, I. D., Kallos, M. S.: Mass transfer limitations in embryoid bodies during human embryonic stem cell differentiation. *Cells Tissues Organs*, **196**, 34–47 (2012).

- Wu, J., Rostami, M. R., Olaya, D. P. C., Tzanakakis, E. S.: Oxygen transport and stem cell aggregation in stirred-suspension bioreactor cultures. *PLoS One*, 9, e102486 (2014).
- Xu, C., Rosler, E., Jiang, J., Lebkowski, J. S., Gold, J. D., O'Sullivan, C., Delavan-Boorsma, K., Mok, M., Bronstein, A., Carpenter, M. K.: Basic fibroblast growth factor supports undifferentiated human embryonic stem cell growth without conditioned medium. *Stem Cells*, 23, 315-323 (2005).
- Xu, R. H., Peck, R. M., Li, D. S., Feng, X., Ludwig, T., Thomson, J. A.: Basic FGF and suppression of BMP signaling sustain undifferentiated proliferation of human ES cells. *Nat. Meth.*, 2, 185-190 (2005).
- Xu, Y., Zhu, X., Hahm, H. S., Wei, W., Hao, E., Hayek, A., Ding, S.: Revealing a core signaling regulatory mechanism for pluripotent stem cell survival and self-renewal by small molecules. *Proc. Natl. Acad. Sci. USA*, **107**, 8129-8134 (2010).
- Zhang, D., Jiang, W., Liu, M., Sui, X., Yin, X., Chen, S., Shi, Y., Deng, H.: Highly efficient differentiation of human ES cells and iPS cells into mature pancreatic insulin-producing cells. *Cell Res.*, **19**, 429–438 (2009).
- Zhang J, Wilson, G. F., Soerens, A. G., Koonce, C. H., Yu, J., Palecek, S. P., Thomson, J. A., Kamp, T. J.: Functional cardiomyocytes derived from human induced pluripotent stem cells. *Circ. Res.*, **104**, e30–41(2009).
- Zweigerdt, R., Olmer, R., Singh, H., Haverich, A., Martin, U.: Scalable expansion of human pluripotent stem cells in suspension culture. *Nat. Protoc.*, **6**, 689-700 (2011).

## **List of Publications**

## **Original Papers:**

- <u>Nath, S.C.</u>, Nagamori, E., Horie, M., Kino-oka, M.: Culture medium refinement by dialysis for the expansion of human induced pluripotent stem cells in suspension culture. *Bioprocess Biosyst. Eng.*, 40, 123-131. doi: 10.1007/s00449-016-1680-z (2016).
- <u>Nath, S.C.</u>, Horie, M., Nagamori, E., Kino-oka, M.: Size- and time-dependent growth properties of human induced pluripotent stem cells in the culture of single aggregate. *J. Biosci. Bioeng.* doi: 10.1016/j.jbiosc.2017.05.006 (in press).

## **International conferences:**

- <u>Nath, S.C.</u>, Kino-oka, M.: High-density culture of human induced pluripotent stem cells through the refinement of medium by dialysis in suspension culture. "Scale-up and manufacturing of cell-based therapies V" conference, San Diego, CA, USA (15<sup>th</sup>-19<sup>th</sup> January, 2017).
- <u>Nath, S.C.</u>, Nagamori, E., Horie, M., Kino-oka, M.: Refining of culture medium by dialysis for the expansion of hiPSC in suspension culture. "The 27<sup>th</sup> annual and international meeting of the Japanese Association for Animal Cell Technology (JAACT)", Kita-Kyushu, Japan (11<sup>th</sup>-14<sup>th</sup> November, 2014).

## Acknowledgements

I would like to express my gratitude to all who make it possible for me to complete this Ph.D. thesis. First of all, I would like to express my deepest gratitude to my supervisor, Professor Masahiro Kino-oka (Department of Biotechnology, Graduate School of Engineering, Osaka University) for giving me the precious opportunity to study in Osaka University, for all of his continuous support, crucial guidance, and valuable advices for my study and life in Japan. With his strict, meticulous, and expert instructions, I have gained and progressed greatly throughout my Doctoral course.

I would also like to express my heart-felt gratitude to Professor Kazuhito Fujiyama (Department of Biotechnology, Graduate School of Engineering, Osaka University), and Professor Takeshi Omasa (Department of Material and Life Science, Graduate School of Engineering, Osaka University) for their kind instructions and valuable advises to complete my Ph.D. thesis perfectly.

I would like to express my sincere thankfulness to Professor Yukako Fujinaga (Faculty of Medicine, Institute of Medical, Pharmaceutical and Health Sciences, Kanazawa University, Kanazawa, Japan) for providing the botulinum hemagglutinin to perform experiments.

I would like to express my sincerest appreciation to my co-supervisor, Assoc. Prof. Eiji Nagamori (Department of Biomedical Engineering, Osaka Institute of Technology, and former Assoc. Prof. in Osaka University) for his guidance throughout the years in various aspects of my research and daily life in Japan.

I would like to express my sincere gratefulness to Assoc. Prof. Mee-Hae Kim (Department of Biotechnology, Graduate School of Engineering, Osaka University) for her unconditional cooperation during my research, original paper preparation, as well as submission. I am also grateful for her kind help during my stay in Kino-oka lab.

I would like to kindly thank Asst. Prof. Masanobu Horie (Division of Biochemical Engineering, Kyoto University, and former postdoc in Kino-oka lab) for providing me the basic training of hiPSCs culture, helping me to prepare original papers, and the best conditions to carry out my research in Kino-oka lab. I have been deeply touched by his philosophy of life and research.

I would like to express my heart-felt gratitude and thankfulness to Ms. Emilda Gomez and Mr. Tokura Tomohiro (ex-Master's course students in Kino-oka lab) for their technical assistance for preparing original paper. I would also like to express my honest thankfulness to Dr. Ngo Xuan Trung for his valuable advises for both life and research. I am also thankful to Dr. Yuuki Ogawa, Dr. Rie Sonoi, Mrs. Menglu Li, and Ms. Xie Yucheng for their continuous support throughout my Doctoral course.

I would also like to express my sincere gratefulness to my current and previous group members, Dr. Hideki Kitajima, Dr. Takaki Ishi, Mr. Motoki Yagi, Mr. Kato Yuma, Mr. Ikata Ryosuke, Mr. Yamamoto Riku, Mr. Kazuhiro Takeuchi, Mr. Ueno Shingo, Ms. Chikayama Yuri, and Mr. Hashida Akihiro, as well as other lab members of Kino-oka lab for their kind cooperation during my experiments, and making my lab-life exciting and unforgettable. I would also like to express my special thanks to all international students of Kino-oka lab for sharing their best times during my stay in Japan.

I also express my gratefulness to Ms. Mitsuko Wakamatsu, Mrs. Kaori Ishida, Mrs. Tomoko Nishimura, and Mrs. Junko Hanashima (secretaries of Kino-oka lab), my Japanese host

115

family (Mrs. Setsu Kawato), and all of my international and Bangladeshi friends for their hearty supports during my stay in Japan.

I would like to express my sincere gratefulness to the organizer of International Program of "Frontier Biotechnology" in the Department of Biotechnology, Graduate School of Engineering, Osaka University for arranging a 5-years course with MEXT Scholarship, and conducting all courses, seminars, and events in English.

I would like to extend my appreciation and gratefulness to the Japanese Ministry of Education, Culture, Sports, Science and Technology for giving me the opportunity to do my Ph. D. as a MEXT scholar at Osaka University, Japan.

Very special thanks goes to my family, especially my wife, Puspa Das for her love, understanding and inspiration, and to my teacher, Assoc. Prof. Imran Hossain (Department of Microbiology, Chittagong University, Bangladesh), who has always encouraged me spiritually to pursue the academic dream!

Nath Suman Chandra Department of Biotechnology Graduate School of Engineering Osaka University Japan July, 2017